

Delft University of Technology

# Stochastic Stability of Discrete-time Phase-coupled Oscillators over Uncertain and Random Networks

Jafarian, Matin; Mamduhi, Mohammad H.; Johansson, Karl H.

DOI 10.1109/TAC.2023.3303102

**Publication date** 2023 **Document Version** Final published version

Published in **IEEE Transactions on Automatic Control** 

# Citation (APA)

Jafarian, M., Mamduhi, M. H., & Johansson, K. H. (2023). Stochastic Stability of Discrete-time Phasecoupled Oscillators over Uncertain and Random Networks. IEEE Transactions on Automatic Control, 69(5), 2915-2930. https://doi.org/10.1109/TAC.2023.3303102

#### Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

#### Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

# Green Open Access added to TU Delft Institutional Repository

# 'You share, we take care!' - Taverne project

https://www.openaccess.nl/en/you-share-we-take-care

Otherwise as indicated in the copyright section: the publisher is the copyright holder of this work and the author uses the Dutch legislation to make this work public.



# Stochastic Stability of Discrete-Time Phase-Coupled Oscillators Over Uncertain and Random Networks

Matin Jafarian<sup>®</sup>, *Member, IEEE*, Mohammad H. Mamduhi<sup>®</sup>, *Senior Member, IEEE*, and Karl H. Johansson<sup>®</sup>, *Fellow, IEEE* 

Abstract—This article studies stochastic relative phase stability, i.e., stochastic phase-cohesiveness, of discretetime phase-coupled oscillators. The stochastic phasecohesiveness in two types of networks is studied. First, we consider oscillators coupled with  $2\pi$ -periodic odd functions over underlying undirected graphs subject to both multiplicative and additive stochastic uncertainties. We prove stochastic phase-cohesiveness of the network with respect to two specific, namely, in-phase and antiphase, sets by deriving sufficient coupling conditions. We show the dependency of these conditions on the size of the mean values of additive and multiplicative uncertainties, as well as the sign of the mean values of multiplicative uncertainties. Furthermore, we discuss the results under a relaxation of the odd property of the coupling function. Second, we study an uncertain network in which the multiplicative uncertainties are governed by the Bernoulli process representing the well-known Erdös-Rényi network. We assume constant exogenous frequencies and derive sufficient conditions for achieving both stochastic phase-cohesive and phase-locked solutions, i.e., stochastic phase-cohesiveness with respect to the origin. For the latter case, where identical exogenous frequencies are assumed, we prove that any positive probability of connectivity leads to phase-locking. Thorough analyses are provided, and insights obtained from stochastic analysis are discussed, along with numerical simulations to validate the analytical results.

Index Terms—Markov processes, nonlinear systems, stochastic systems, synchronization of coupled oscillators.

Manuscript received 21 April 2023; accepted 17 July 2023. Date of publication 7 August 2023; date of current version 26 April 2024. The work of Matin Jafarian was supported by the Horizon 2020 Marie-Curie Fellowship, Project ReWoMen under Grant 101030415. The work of Karl H. Johansson was supported in part by the Knut and Alice Wallenberg Foundation and in part by the Swedish Research Council. Recommended by Associate Editor M. Cao. (*Corresponding author: Matin Jafarian.*)

Matin Jafarian is with the Delft Center for Systems and Control, Delft University of Technology, 2628 CD Delft, Netherlands (e-mail: m.jafarian@tudelft.nl).

Mohammad H. Mamduhi was with the Automatic Control Laboratory, ETH Zürich, 8092 Zürich, Switzerland (e-mail: mamduhi@kth.se).

Karl H. Johansson is with the Division of Decision and Control Systems, School of Electrical Engineering and Computer Science, KTH Royal Institute of Technology, 11428 Stockholm, Sweden, and also with the Digital Futures, 10044 Stockholm, Sweden (e-mail: kallej@kth.se).

Digital Object Identifier 10.1109/TAC.2023.3303102

#### I. INTRODUCTION

O SCILLATORY behavior is a fundamental feature of complex networks in a wide range of applications from mechanical and electrical power networks to biological and neuronal networks [1], [2], [3], [4], [5], [6]. Synchronization is useful for achieving stable oscillations in these networks [2]. Uncertainties and randomness influence the ability of a network to synchronize. A prominent example is a network of biological neurons whose activities govern the underlying mechanisms of cognition [1] and motion [7]. In such neuronal networks, the coupling and the input current of neurons are often subject to fluctuations due to various stochastic uncertainties [8], [9], [10], [11]. Therefore, studying the stability of stochastic uncertain oscillatory models is essential.

Analyzing the behavior of coupled oscillators is hard even in the absence of uncertainties, due to the nonlinear nature of their dynamics. One approach to reduce the complexity of analyses is to map the oscillators' models to their phase dynamics. Such a transformation is feasible under particular conditions [12], [13], [14]. The framework has been employed in studying synchronization of some classes of nonlinear coupled oscillators, for instance, van der Pol oscillators, and weakly coupled neurons [1], [15], [16]. Among notions of synchronization is phase-cohesiveness, a principal desired behavior of oscillatory networks implying boundedness of relative phases [17]. Phasecohesiveness extends the notion of phase-locking (also called phase-synchronization), and is useful in achieving frequency synchronization [17]. Despite the importance of stochastic synchronization, thus far, phase-cohesiveness in stochastic models has not been studied in literature. Our objective is to address this gap by studying the effects of stochastic uncertainties on phase-coupled oscillators.

In phase-coupled models, oscillators are interconnected via nonlinear periodic functions. The most studied model of phasecoupled oscillators is the well-known Kuramoto model [3], in which the coupling law is reduced to its leading term in an odd Fourier expansion, i.e., sine function. The continuous-time deterministic Kuramoto model has been widely employed to study synchronization in various applications [17], [18], [19], [20], [21], [22]. Besides, the importance of more general classes of coupling functions on synchronization has also been shown in literature (see [15], [23], [24], [25]).

We study the effects of stochastic multiplicative and additive uncertainties in a network of phase oscillators coupled via  $2\pi$ -periodic odd functions. Odd coupling functions generalize the Kuramoto model, by allowing the addition of some higher

1558-2523 © 2023 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See https://www.ieee.org/publications/rights/index.html for more information.

Fourier harmonics (e.g., phase model of the locus coeruleus neurons [15]), and present a gradient dynamical system, which has been proven useful in achieving convergence in oscillatory networks [16], [26], [27]. For instance, phase oscillators coupled via odd functions have been used in studying the associate memory feature of integrate-and-fire spiking neuronal networks [16], as well as recall of oscillatory associative memory networks [25]. The combination of generality, applicability, and mathematical amenability inspire us to primarily focus on this class of coupling functions. Furthermore, we discuss an extension of our results by allowing a relaxation of the odd property of the coupling function.

This article studies stochastic set stability for phase oscillators described above, using a discrete-time approximation, and perform the analyses within the framework of Markov chain stability [28], [29].

#### A. Related Works

The effects of uncertainties on the synchronization of oscillatory networks have mainly been investigated in continuous-time deterministic perturbed models, and stochastic models with only additive noise. In the stochastic setting, the effects of additive noise, such as uncertain exogenous frequencies, have been studied for some classes of continuous-time nonlinear oscillators using a phase reduction model [30], and a Fokker–Planck model [26], [31]. Moreover, synchronization of the continuoustime Kuramoto oscillators with additive uncertainties, modeled as a Wiener process, has been analyzed in a stochastic game setting [32]. A mean-filed approach to frequency synchronization of the continuous-time coupled Kuramoto oscillators subject to random exogenous frequencies has been studied in [33]. Oscillators in the Erdös–Rényi [34], [35] random networks, in which each two oscillators are decoupled with a nonzero probability, have also been studied [36], [37], [38], [39]. All aforementioned studies have focused on the asymptotic full-state synchronization of continuous-time oscillators in the Euclidean space using a linear approximation of the network dynamics in the vicinity of the synchronous state.

The effects of coupling noise have been studied on the fullstate synchronization of a network of continuous-time harmonic oscillators [40], and interconnected oscillators with a common or zero intrinsic noise together with state-dependent coupling noise [41], [42]. These works have analyzed the network behavior using stochastic differential equations and discussed the effects of noise on the synchronization.

Deterministic perturbed Kuramoto model has been studied considering time-varying exogenous frequencies and coupling coefficients [43], [44], [45]. The local input-output stability of the exact synchronization (phase-locking) solution has been proven imposing an upper-bound on the time-varying exogenous frequencies [43]. Moreover, phase-cohesiveness of the in-phase solution in a network with time-varying couplings and exogenous frequencies [44] has been shown imposing conditions on the time-evolution of the coupling coefficients. Compared with literature, our contributions, detailed in the following, include: studying stochastic phase-coupled oscillators; generalizing the coupling function; considering stochastic multiplicative and additive uncertainties (see Remark 1); and studying the stability of phase-coupled oscillators in random networks. It is worth noting that the deterministic unperturbed discrete-time Kuramoto oscillators, with applications in communication and robotics networks, have also been studied [46], [47]. Our construction of the Lyapunov function for the general connected and undirected networks, extends the analysis of the aforementioned works, if we replace the stochastic variables with the deterministic ones.

#### B. Main Contributions

Our objective is to study the effects of multiplicative, i.e., system's states or a function of them multiplied by stochastic random variables [48], and additive uncertainties on the stability of relative phases of discrete-time phase-coupled oscillators. Our main contributions are highlighted as follows. First, we introduce the new notions of stochastic phase-cohesiveness and ultimate stochastic phase-cohesiveness. The notion of phasecohesiveness for deterministic models is defined based on the concept of invariant sets [17]. For stochastic models, we use the concept of Harris recurrent Markov chains. Basically, a network of stochastic phase-coupled oscillators is phase-cohesive with respect to a desired set if the probability that the relative phases return to this set, after leaving it, is one (see Definition 6). Ultimate stochastic phase-cohesiveness indicates a bounded return time (see Definition 7). Second, we study the phase-cohesiveness of discrete-time phase oscillators over an undirected graph subject to both multiplicative (i.e., coupling weights) and additive (i.e., exogenous frequencies) stochastic uncertainties. The presence of multiplicative uncertainties indicates that the underlying topology is connected in a probabilistic sense. We consider  $2\pi$ -periodic, bounded, and odd coupling laws, and independent and identically distributed (i.i.d) random uncertainties obeying normal distributions. In Theorem 1, we assume either strictly positive or negative mean values for the uncertain couplings. By obtaining sufficient conditions in the form of lower bounds on the common coupling coefficient, we prove that depending on the sign of the mean values of multiplicative uncertainties, the uncertain network achieves stochastic phase-cohesiveness with respect to either in-phase or antiphase sets. These sets are defined in the vicinity of the roots of the coupling function, i.e., zero (in-phase) and  $\pi$  (antiphase). We also characterize the conditions for achieving ultimate phase-cohesiveness (see Corollary 1). Respecting the discrete-time setting, our results are derived assuming a sufficiently small sampling-time for which we also characterize an upper bound.

In Proposition 1, we then allow the coexistence of edges whose corresponding multiplicative uncertainties possess either negative or positive mean values. Assuming identical exogenous frequencies, we study this special case over an underlying line topology. The conditions under which the network exhibits clustering behavior are derived by showing its stochastic phasecohesiveness with respect to the union of the in-phase and antiphase sets.

Third, we discuss a relaxation of the odd property of the coupling function. We allow the coupling function to be non-odd only on a subset of its domain. This relaxation describes a more general coupling law which is applicable, for instance, in studying series arrays of the Josephson junctions [24]. For oscillators over an undirected tree network with multiplicative (positive mean-values) and additive uncertainties, conditions under which the network is stochastic phase-cohesive with respect to the in-phase set are characterized (see Proposition 2). Fourth, we study phase-coupled oscillators, with  $2\pi$ -periodic odd couplings, such that each two coupled oscillators are connected using a common and constant coupling term and with a non-zero

probability of connection, which represents an Erdös-Rényi random network. We obtain conditions for achieving both stochastic phase-cohesive and phase-locked solutions for this network. For the case of constant and non-identical frequencies, we show that the effect of the randomness is to stabilize the in-phase set as the only absorbing set of the network (see Theorem 2). Furthermore, we prove that any positive probability of connection will lead to a phase-locking solution when all oscillators have identical exogenous frequencies (see Corollary 2).

In Remark 1, stochastic analysis is compared with deterministic perturbation analysis. To the best of our knowledge, the stochastic stability of phase-coupled oscillators has not been studied in literature. A preliminary result for the Kuramoto oscillators over a tree network was presented in our conference paper [49].

#### C. Outline

The rest of this article is organized as follows. Section II provides the model and the required preliminaries. Section III presents the problem formulations, and the notions of stochastic phase-cohesiveness, and ultimate stochastic phasecohesiveness. The analyses of uncertain and random networks are presented in Sections IV and V, respectively. Section VI presents simulation results. Finally, Section VII concludes this article.

#### D. Notation

The notation  $x_{i,j}$  is equivalently used for  $x_i - x_j$ . A random variable x selected from an arbitrary distribution  $\mathcal{X}$  with mean  $\mu$  and variance  $\sigma^2$  is denoted by  $x \sim \mathcal{X}(\mu, \sigma^2)$ . The expected value and conditional expected value operators are denoted by  $E[\cdot]$  and  $E[\cdot|\cdot]$ , respectively. The symbol  $\mathbb{S}^1$  denotes the unit circle. An angle is a point in  $\mathbb{S}^1$  and an arc is a connected subset of  $\mathbb{S}^1$ . Given a matrix M of real numbers, we denote by  $\mathcal{R}(M)$ and  $\mathcal{N}(M)$  the range and the null space, respectively. Symbol  $\mathbf{1}_n$ is an n-dimensional vector of all ones. The empty set is denoted by ∅.

#### II. MODEL AND PRELIMINARIES

This article considers a network of discrete-time phasecoupled oscillators governed by dynamics in (1), which is a discrete-time approximation of its continuous-time counterpart

$$\theta_{i}(\mathbf{k}+1) = \left\{ \theta_{i}(\mathbf{k}) + \tau \tilde{\omega}_{i}(\mathbf{k}) - \left( \kappa \tau \sum_{j \in \mathcal{N}_{i}} \tilde{\alpha}_{i,j}(\mathbf{k}) \Psi(\theta_{i,j}(\mathbf{k})) \right) \right\} \pmod{2\pi}$$
(1)

where  $\mathbf{k} \in \mathbb{Z}^+$ ,  $\theta_i(\mathbf{k}) \in \mathbb{S}^1$ ,  $\theta_{i,j}(\mathbf{k})$ ,  $\tau > 0$ , and  $\kappa > 0$  represent the time step, the phase of oscillator i, the relative phase of oscillators i and j, the sampling time, and the common coupling term, respectively. The set of neighbors of oscillator i is denoted by  $\mathcal{N}_i$ . The variables  $\tilde{\alpha}_{i,j}(\mathbf{k})$  and  $\tilde{\omega}_i(\mathbf{k})$  are stochastic variables representing the uncertain multiplicative coupling weight, and the uncertain exogenous frequency, respectively. Function  $\Psi(\cdot)$ is the coupling function. For the purpose of brevity, the term

 $(\mod 2\pi)$  is omitted from all representations of the discretetime dynamics in the rest of this article.

Assumption 1: Function  $\Psi(\cdot)$  is  $2\pi$ -periodic, continuously differentiable, and odd.<sup>1</sup> In the interval  $[0, \pi]$ , it holds that  $\Psi(0) = \Psi(\pi) = 0$ , and  $|\Psi(\xi)| \neq 0 \forall \xi \in (0, \pi)$ .

Since  $\Psi(\cdot)$  is continuously differentiable, hence bounded, i.e.,  $|\Psi(\xi)| \leq \Psi_{\text{max}}$ , the following property immediately follows. This property indicates that there exits an arc  $\Upsilon \subset [0, \pi]$  such that the value of  $\Psi(\cdot)$  for every angle on this arc is greater than angles which belong to arcs in the vicinity of 0 and  $\pi$ .

*Property 1:* For any function  $\Psi(\cdot)$  satisfying Assumption 1, there exists an arc  $\Upsilon = (\gamma, \gamma_{\max}) \subset (0, \pi)$ , such that  $\forall \xi, z \in$  $[0,\pi]$ , if  $\xi \notin \Upsilon$ ,  $z \in \Upsilon$  then  $|\Psi(\xi)| \leq |\Psi(z)|$ .

#### A. Preliminaries

The term  $|\theta_i(\mathbf{k}) - \theta_i(\mathbf{k})|$  denotes the geodesic distance between phases  $\theta_i, \theta_i \in \mathbb{S}^1$ . The geodesic distance is the minimum value between the counter-clockwise and the clockwise arc lengths connecting  $\theta_i$  and  $\theta_j$ . The size of the relative phase  $\theta_{i,j} = \theta_i - \theta_j \in (-\pi, \pi]$  equals  $|\theta_i - \theta_j|$  and its sign is positive if the counter-clockwise path length from  $\theta_i$  to  $\theta_j$  is smaller than that of the clockwise path. The relative phase of the two oscillators decreases if  $|\theta_{i,j}(\mathbf{k}+1)| < |\theta_{i,j}(\mathbf{k})|$ .

#### A. Graph Theory

We here revisit some preliminaries on graph theory mainly borrowed from [50]. Consider an undirected graph  $G(\mathcal{V}, \mathcal{E})$ , where  $\mathcal{V}$  is the set of *n* nodes and  $\mathcal{E} \subset \mathcal{V} \times \mathcal{V}$  is the set of m edges. The graph's incidence matrix is denoted by  $B_{n \times m}$ . The two matrices  $L(G) \triangleq BB^T$  and  $L_e(G) \triangleq B^T B$  are called the graph Laplacian and the edge Laplacian, respectively. If the underlying graph is connected, then the eigenvalues of L can be ordered as  $0 = \lambda_1(L) < \lambda_2(L) \leq \ldots \leq \lambda_n(L)$ , where  $\lambda_2(L)$  is called the algebraic connectivity of the graph. Moreover, all nonzero eigenvalues of  $L_e$  are equal to the nonzero eigenvalues of L.

A spanning tree of G is a subgraph  $G_{\tau}(\mathcal{V}, \mathcal{E}_{\tau})$  which is a tree (cycle free) graph. Under an appropriate permutation of the edge indexes, the incidence matrix of a connected graph G can be partitioned as  $B = [B(G_{\tau}) \ B(G_{c})]$ , where  $G_{\tau}$  represents a given spanning tree of G, and  $G_c$  represents the remaining edges. There exists a matrix R such that  $L_e(G) = R^{\top} L_e(G_{\tau}) R$ , where  $R = \begin{bmatrix} I & T \end{bmatrix}$  with  $T = L_e^{-1}(G)B^{\top}(G_{\tau})B(G_c)$ . For tree graphs, the edge Laplacian is positive definite.

#### A. Markov Chains

A general measurable space is a pair  $(X, \mathcal{B}(X))$  with X a set of points and  $\mathcal{B}(X)$  a  $\sigma$ -algebra of subsets of X satisfying the following properties.

1)  $\emptyset \in \mathcal{B}(X).$ 

2) If  $D \in \mathcal{B}(X)$ , then  $D^c \in \mathcal{B}(X)$ , where  $D^c = X \setminus D$ .

3) If  $D_1 \in \mathcal{B}(X)$  and  $D_2 \in \mathcal{B}(X)$ , then  $D_1 \cup D_2 \in \mathcal{B}(X)$ .

A Markov chain is a stochastic process  $\Phi = \{\Phi_0, \Phi_1, \ldots\}$ such that each  $\Phi_i$  is randomly taking values on the measurable state space X which is endowed by the  $\sigma$ -algebra  $\mathcal{B}(X)$ . The chain  $\Phi$  is defined by the triple  $(\Omega, \mathcal{F}, \mathsf{P})$ , such that:

<sup>&</sup>lt;sup>1</sup>The relaxation of this assumption is discussed in Section IV.

- 1)  $\Omega$  is the whole state space, i.e., the product of all pairs, each of which corresponds to  $\Phi_i$  and is a subset of  $(X, \mathcal{B}(X))$ ;
- 2)  $\mathcal{F}$  is a  $\sigma$ -algebra associated to the measurable space  $\Omega$ ;
- P: F→[0,1] is a probability measure defined on (Ω, F) that assigns a probability to each outcome of F.

The following definitions are mainly borrowed from [29].

Definition 1: [29, Ch. 3] Let  $(X, \mathcal{B}(X))$  be a measurable space. The state space X is called *countable* if X is discrete, with a countable number of elements, and with  $\mathcal{B}(X)$  the  $\sigma$ -algebra of all subsets of X. The state space X is called *general* if it is assigned a countably *generated*  $\sigma$ -algebra  $\mathcal{B}(X)$ .<sup>2</sup>

Definition 2: For a stochastic process  $\Phi = \{\Phi_0, \Phi_1, ...\}$  defined on  $(\Omega, \mathcal{F}, \mathsf{P})$ , let  $P^n(\omega, \mathcal{B})$  denote the transition probability that the process enters the set  $\mathcal{B}$  after n transitions, *i.e.*,  $\Phi_{n+m} \in \mathcal{B}$ , given  $\Phi_m = \omega$ . Then  $\Phi$  is a *time-homogeneous* Markov chain if transition probabilities  $\{P^n(\omega, \mathcal{B}), \omega \in \Omega, \mathcal{B} \subset \Omega\}$  exist, such that for any  $n, m \in \mathbb{Z}^+$ , the following holds:

$$\mathsf{P}(\Phi_{n+m} \in \mathcal{B} | \Phi_j, j \le m, \Phi_m = \omega) = P^n(\omega, \mathcal{B}).$$

The independence of the transition probability  $P^n(\omega, \mathcal{B})$  from  $j \leq m$  entails the Markov property, and its independence from m confirms the time-homogeneity.

Definition 3: Let  $\Phi = \{\Phi_0, \Phi_1, \ldots\}$  be a Markov chain defined on  $(\Omega, \mathcal{F}, \mathsf{P})$ . Then:

1) for any  $D \in \mathcal{F}$ , the measurable function  $\tau_D : \Omega \to \mathbb{Z}^+ \cup \{\infty\}$  is the *first return time* to the set D, i.e.,

$$\tau_D \triangleq \min\{n \ge 1 \mid \Phi_n \in D\} \tag{2}$$

for any measure φ on the σ-algebra F, the Markov chain Φ is said to be φ-*irreducible* if ∀ω ∈ Ω and D ∈ F, φ(D) > 0 implies P(τ<sub>D</sub> < ∞) > 0.

According to Definition 3, the entire state space of a Markov chain is reachable, independent of the initial state, via finite number of transitions if the Markov chain is  $\varphi$ -irreducible. Moreover, in that case, a unique maximal irreducibility measure  $\psi > \varphi$  exists on  $\mathcal{F}$  such that  $\Phi$  is  $\varphi'$ -irreducible for any other measure  $\varphi'$  if and only if  $\psi > \varphi'$ . We then say that the Markov chain is  $\psi$ -irreducible.

Definition 4: [51, Ch. 5] Let  $a = \{a(n)\}$  be a probability measure on  $\mathbb{Z}^+$  and  $\Phi_a$  be the sampled chain of the Markov chain  $\Phi$  at time-points drawn successively according to the distribution a. Denote the probability transition kernel of  $\Phi_a$ by  $K_a(\omega, D) = \sum_{n=0}^{\infty} K^n(\omega, D) a(n)$ , where  $K(\omega, D)$  and  $K^n(\omega, D)$  are the probability transition kernel of  $\Phi$ , and the *n*-step probability transition kernel of  $\Phi$ , respectively. A set  $C \in \mathcal{B}$  is a  $\nu_a$ -petite set, where  $\nu_a$  is a nontrivial measure on  $\mathcal{B}$ , if for all  $\omega \in C$  and  $D \in \mathcal{B}$ ,  $\Phi_a$  satisfies

$$K_a(\omega, D) \ge \nu_a(D).$$

Definition 5. [51, Ch. 6]: Let K be the transition probability kernel of a chain  $\Phi$  defined on a locally compact and separable space X acting on a bounded function  $h(x) : X \to \mathbb{R}, x \in X$ via the mapping  $Kh(x) = \int K(x, dy)h(y)$ . Denoting the class of bounded continuous functions from X to  $\mathbb{R}$  by  $\mathcal{C}(X)$ , the chain  $\Phi$  has the Feller property if K maps  $\mathcal{C}(X)$  to  $\mathcal{C}(X)$ .

TABLE I MOST USED VARIABLES AND NOTATIONS

$\theta_i(\mathbf{k})$	Phase of oscillator $i$ at time-step k
$\theta_{i,j}(\mathbf{k})$	Relative phase, <i>i.e.</i> , $\theta_i(\mathbf{k}) = \theta_j(\mathbf{k})$
$\boldsymbol{ heta}(\mathbf{k})$	Augmented phase vector
$\Theta(\mathbf{k})$	Augmented relative phase vector
$\mathbb{S}^1$	Unit circle
$\kappa$	Common coupling strength
au	Common sampling time
$\Psi$	Periodic odd coupling function
$\tilde{\alpha}_{i,j}(\mathbf{k})$	Multiplicative uncertainty of edge $(i, j)$
$ ilde{\omega}_i({ m k})$	Uncertain exogenous freq. of oscillator $i$
${\mathcal E}$	Edge set of the underlying network graph
$\mathcal{N}_{i}$	Set of neighbors of oscillator $i$
Π	State space of augmented relative phases
$\gamma$	Phase angle on the arc $[0, \pi]$
$S_G^{\Psi}(\gamma)$	In-phase set
$\tau_{S_C^{\Psi}(\gamma)}$	First return time to in-phase set
$S_C^{\Psi}(\gamma_{\max})$	Anti-phase set

#### III. PROBLEM FORMULATION

This section presents the definition of the stochastic phasecohesiveness, and presents the problem statement. A list of mostly used variables is summarized in Table 1.

#### A. Stochastic Phase-Cohesiveness

This section introduces two new definitions for stochastic stability of phase-coupled oscillators based on two notions of stability of stochastic processes. Our first definition, stochastic phase-cohesiveness, corresponds to the concept of Harris recurrent Markov chains [29, Ch. 9]. A  $\psi$ -irreducible chain with state space X is Harris recurrent if it visits every set  $A \in \mathcal{B}^+(X)$  almost infinitely, where  $\mathcal{B}^+(X) \triangleq \{A \in \mathcal{B}(X) : \psi(A) > 0\}$  and  $\psi$  is the maximal irreducibility measure.<sup>3</sup> Equivalently, a chain is Harris recurrent if the probability of its first return time  $\tau_A$  to a desired set A is one [29].

Let  $\Theta$  denote the augmented relative phase vector of the interconnected oscillators with the dynamics in (1)

$$\Theta(\mathbf{k}) = \mathcal{H}^G(\boldsymbol{\theta}(\mathbf{k})) \tag{3}$$

where  $\boldsymbol{\theta}(\mathbf{k}) \triangleq [\theta_1(\mathbf{k}), \dots, \theta_n(\mathbf{k})]^\top$  is the augmented phase state, and  $\mathcal{H}^G$  is a topology-dependent operator which computes the relative phases. The state space on which  $\Theta(\mathbf{k})$  evolves is defined by

$$\Pi = [-\pi, \pi]^{\|\mathcal{E}\|} \tag{4}$$

where  $||\mathcal{E}||$  denotes the number of edges of the underlying network topology.

Definition 6 (Stochastic phase-cohesiveness): The relative phase process  $\Theta(\mathbf{k})$  in (3) is stochastic phase-cohesive if  $\mathsf{P}_{\Theta(\mathbf{k})}(\tau_{S_{G}^{\Psi}(\gamma)} < \infty) = 1$ , where  $S_{G}^{\Psi}(\gamma) \subset \Pi$  is a desired compact set, and  $\tau_{S_{G}^{\Psi}(\gamma)}$  is the first return time of the stochastic process  $\Theta(\mathbf{k})$  to the set  $S_{G}^{\Psi}(\gamma)$ .

The above definition requires that the Markov chain returns to a desired set  $S_G^{\Psi}(\gamma)$  almost infinitely. In what follows, we present

<sup>&</sup>lt;sup>2</sup>The smallest  $\sigma$ -algebra on which  $\mathcal{B}$  is measurable, i.e., the intersection of all  $\sigma$ -algebras on which  $\mathcal{B}$  is measurable, is called the generated  $\sigma$ -algebra by  $\mathcal{B}$ .

<sup>&</sup>lt;sup>3</sup>The set  $\mathcal{B}^+(X)$  contains all sets of positive  $\psi$ -measure subsets of  $\mathcal{B}(X)$  and is uniquely defined for  $\psi$ -irreducible chains [29, Ch.4].



Fig. 1. Relative phase of two oscillators i and j and its map on the arcs  $\underline{\Upsilon}, \Upsilon$ , and  $\overline{\Upsilon}$ .

a stronger notion based on the concept of positive Harris recurrent chains [29]. Conceptually, the following definition is the stochastic counterpart of deterministic ultimate boundedness. It implies that not only the chain revisits  $S_G^{\Psi}(\gamma)$  almost infinitely but also the return time is bounded and the chain probability distribution converges to an invariant probability measure in a stationary regime.

Definition 7 (Ultimate stochastic phase-cohesiveness): The relative phase process  $\Theta(\mathbf{k})$  in (3) is ultimate stochastic phase-cohesive if there exists a constant  $M < \infty$  such that  $\sup \mathbf{E}_x[\tau_{S_G^{\Psi}(\gamma)}] \leq M$  for all  $x \in S_G^{\Psi}(\gamma)$ , where  $\mathbf{E}_x$  denotes the expectation of events conditional on the chain beginning with  $\Phi_0 = x$ .

#### B. In-Phase and Antiphase Sets

Here, we introduce two sets, namely, in-phase and antiphase, which are useful in our problem statement. For a function  $\Psi(\cdot)$ , satisfying Assumption 1 and Property 1, choose  $\gamma$  and  $\gamma_{\max}$  such that  $0 < \gamma < \gamma_{\max} < \pi, 0 < |\Psi(\gamma)| = |\Psi(\gamma_{\max})| < \Psi_{\max}$  hold. Define the arcs  $\underline{\Upsilon}$ ,  $\Upsilon$ , and  $\overline{\Upsilon}$  as follows:

$$\underline{\Upsilon} = [0, \gamma] : \forall \gamma_i \in \underline{\Upsilon} : 0 \le |\Psi(\gamma_i)| \le |\Psi(\gamma)|$$
(5)

$$\Upsilon = (\gamma, \gamma_{\max}) : \forall \gamma_i \in \Upsilon : |\Psi(\gamma)| < |\Psi(\gamma_i)| < \Psi_{\max}$$
(6)

$$\overline{\Upsilon} = [\gamma_{\max}, \pi] : \forall \gamma_i \in \overline{\Upsilon} : 0 \le |\Psi(\gamma_i)| \le |\Psi(\gamma)|.$$
(7)

Fig. 1 depicts the introduced arcs together with the relative phase of two oscillators i and j.

We introduce the in-phase set  $S_G^{\Psi}(\gamma) \subset \Pi$  and antiphase  $S_G^{\Psi}(\gamma_{\max}) \subset \Pi$ , as

$$S_{G}^{\Psi}(\gamma) = \{\theta_{i} \in \mathbb{S}^{1}, \theta_{j} \in \mathbb{S}^{1} : |\theta_{i,j}(\mathbf{k})| \in \underline{\Upsilon}, \forall (i,j) \in \mathcal{E}\}$$
(8)

$$S_G^{\Psi}(\gamma_{\max}) = \{ \theta_i \in \mathbb{S}^1, \theta_j \in \mathbb{S}^1 \colon |\theta_{i,j}(\mathbf{k})| \in \overline{\Upsilon} \forall (i,j) \in \mathcal{E} \}.$$
(9)

#### C. Problem Statement

Here, we formulate the two problems to be studied in Sections IV and V, respectively.

*Problem 1:* Consider a network of n phase-coupled oscillators, where the dynamics of each oscillator obeys (1). Let

$$\tilde{\alpha}_{i,j}(\mathbf{k}) = \alpha_{i,j} + n_{i,j}(\mathbf{k}), \quad \tilde{\omega}_i(\mathbf{k}) = \omega_i + \varpi_i(\mathbf{k})$$
(10)

where  $\alpha_{i,j}, \omega_i \in \mathbb{R}^+$  are constants,  $n_{i,j}(\mathbf{k})$  and  $\overline{\omega}_i(\mathbf{k})$  are i.i.d. stochastic variables at each time step  $\mathbf{k}$  selected from a continuous distribution with finite mean and variance. Also, assume that the initial relative phase  $\theta_i(0) - \theta_j(0)$  is an arbitrary random variable, independent from the realizations of  $n_{i,j}(\mathbf{k})$  and  $\varpi_i(\mathbf{k})$ ,  $\forall \mathbf{k}$ , and  $\forall i, j$ , with a finite moment probability distribution with a continuous density function.

Denote the relative phase stochastic process corresponding to this network by  $\Theta(k)$ , as in (3). Our objective is to study the stability of  $\Theta(k)$ , especially, to characterize coupling,  $\kappa$ , condition under which the process is stochastic phase-cohesive with respect to the in-phase and antiphase sets defined in (8) and (9).

Problem 2: Now, consider a special case of the network in Problem 1 by assuming that the multiplicative uncertainties obey the Bernoulli distribution. That is, two oscillators *i* and *j* are coupled with probability *p* and decoupled with probability 1 - p. When two oscillators are coupled, the weight of their corresponding edge equals  $\kappa > 0$ . We assume constant exogenous frequencies. To make a distinction with Problem 1, denote the multiplicative uncertainty by  $\beta_{i,j}$ . The dynamics of oscillator *i* can then be expressed as

$$\theta_i(\mathbf{k}+1) = \theta_i(\mathbf{k}) - \left(\kappa\tau \sum_{j\in\mathcal{N}_i} \beta_{i,j}(\mathbf{k})\Psi(\theta_{i,j}(\mathbf{k}))\right) + \tau\omega_i$$
(11)

where  $\omega_i \in \mathbb{R}^+$  represents the constant exogenous frequency of oscillator *i*.

Our objective is to study the stability of this random network and obtain coupling conditions under which the relative phase stochastic process  $\Theta(k)$  is stochastic phase-cohesive with respect to the in-phase set. Moreover, we study conditions under which the phase-locked solution, i.e., phase-cohesiveness with respect to  $S_G^{\Psi}(0)$ , is achieved.

#### **IV. OSCILLATORS IN AN UNCERTAIN NETWORK**

This section studies Problem 1 to characterize conditions under which stochastic and ultimate stochastic phase-cohesiveness are achieved for the network.

Considering the dynamics of individual oscillators in (1) and stochastic variables in (10), the relative phase dynamics of two interconnected oscillators i and j follows:

$$\theta_{i,j}(\mathbf{k}+1) = \theta_{i,j}(\mathbf{k}) + \tau \tilde{\omega}_{i,j}(\mathbf{k}) - \left(\kappa \tau \sum_{\ell \in \{i,j\}} \sum_{e \in \mathcal{N}_{\ell}} \tilde{\alpha}_{\ell,e}(\mathbf{k}) \Psi(\theta_{\ell,e}(\mathbf{k}))\right).$$
(12)

Assumption 2: The underlying deterministic network topology  $G(\mathcal{V}, \mathcal{E})$  is connected. The uncertain interconnections of oscillators are undirected, i.e.,  $\tilde{\alpha}_{i,j}(\mathbf{k}) = \tilde{\alpha}_{j,i}(\mathbf{k})$ . Each multiplicative random variable obeys the normal distribution, i.e.,  $\tilde{\alpha}_{i,j} \sim \mathcal{N}(\mu_{i,j}, \sigma^2)$ . For each of the nominal exogenous frequencies, it holds that  $\boldsymbol{E}[\tilde{\omega}_i] > 0$ .

Notice that identical variances for multiplicative uncertainties are not needed in our analysis. This assumption is only made for the sake of clarity of representation (see Theorem 1). From Assumption 2, the relative phase vector for the whole network,  $\Theta(k)$  in (3), is equal to  $B^{\top}\theta(k)$ , where B is the incidence matrix of the underlying deterministic graph  $G(\mathcal{V}, \mathcal{E})$ . The augmented relative phase dynamics can then be written in the following compact form:

$$B^{\top}\boldsymbol{\theta}(\mathbf{k}+1) = B^{\top} \left(\boldsymbol{\theta}(\mathbf{k}) + \tau \tilde{\boldsymbol{\omega}}(\mathbf{k}) - \tau \kappa B \boldsymbol{\alpha}(\mathbf{k}) \Psi(B^{\top}\boldsymbol{\theta}(\mathbf{k}))\right)$$
(13)

$$\tilde{\boldsymbol{\omega}}_{n \times 1}(\mathbf{k}) = (\tilde{\omega}_1(\mathbf{k}), \dots, \tilde{\omega}_n(\mathbf{k}))^{\top}$$

$$\mathbf{\tilde{\alpha}}_{m \times m}(\mathbf{k}) = \begin{pmatrix} \tilde{\alpha}_{l_1}(\mathbf{k}) & 0 & \dots & 0\\ 0 & \tilde{\alpha}_{l_2}(\mathbf{k}) & \dots & 0\\ \vdots & \vdots & \ddots & \vdots\\ 0 & 0 & \dots & \tilde{\alpha}_{l_m}(\mathbf{k}) \end{pmatrix}$$

where  $l_p \in \mathcal{E}$ , with  $p \in \{1, ..., m\}$ , denotes the *p*th edge of the underlying graph  $G(\mathcal{V}, \mathcal{E})$ .

#### A. Markov Properties

Before studying the network's behavior, we discuss essential properties of the stochastic process described in (13).

*Lemma 1:* The relative phase stochastic process (13) is a timehomogeneous and  $\psi$ -irreducible Markov chain evolving in a general space.

The above properties guarantee that a nonzero probability exists for the Markov chain to make a transition from any initial state to any state in the whole state space. In the following result, we show that every compact set in the state space  $\Pi$  is also a petite set (Definition 4). This equivalence is indeed helpful in studying stochastic stability of the Markov chain (13), as will be shown in Section IV-B.

*Lemma 2:* Every compact set in the state space of the relative phase stochastic process (13) is a petite set.

#### B. Stochastic Phase-Cohesiveness

In this section, we study the phase-cohesiveness of the coupled oscillators modeled in (13) employing the mathematical tools for the stability analysis of Markov chains. We first derive sufficient conditions under which the relative phase Markov chain is Harris recurrent and hence stochastic phase-cohesive. Based on [29, Th. 9.1.8], a  $\psi$ -irreducible chain  $\Phi$  defined on a state space X is Harris recurrent if there exists a petite set  $C \subset X$ , and a function  $V : X \to \mathbb{R}^+$ , which is unbounded off petite sets (i.e., all sublevel sets of V are petite), such that the following drift condition is satisfied:

$$\Delta V = \boldsymbol{E}[V(\Phi_{k+1})|\Phi_k = x] - V(x) < 0 \quad \forall \Phi_k \in X \setminus C.$$
(14)

Our results are based on an application of the mentioned theorem for our network problem setting. For the purpose of illustration, we first provide an example by analyzing the behavior of a network of two coupled oscillators.

*Example 1:* Consider a network of two oscillators modeled as the discrete-time Markov chain in (13)

$$\theta_{1,2}(\mathbf{k}+1) = \underbrace{\theta_{1,2}(\mathbf{k}) - 2\tau\kappa \,\tilde{\alpha}_{1,2}(\mathbf{k})\Psi(\theta_{1,2}(\mathbf{k}))}_{a} + \tau \underbrace{\tilde{\omega}_{1,2}(\mathbf{k})}_{b}$$

with  $\theta_{1,2} = \theta_1 - \theta_2$ ,  $\tilde{\omega}_{1,2} = \tilde{\omega}_1 - \tilde{\omega}_2$ , and  $\tilde{\alpha}_{1,2} \sim N(\mu, \sigma^2)$ . Define  $\underline{\Upsilon} = [0, \gamma], \Upsilon = (\gamma, \gamma_{\max}), \overline{\Upsilon} = [\gamma_{\max}, \pi]$ . We will show that  $S_G^{\Psi}(\gamma) = \{\theta_i \in \mathbb{S}^1, \theta_j \in \mathbb{S}^1 : |\theta_{1,2}| \in \underline{\Upsilon}\}$  is an absorbing set for the chain, *i.e.*, the chain is Harris recurrent with respect to this set. We will further show that the chain is transient on  $S_G^{\Psi}(\gamma_{\max}) = \{\theta_i \in \mathbb{S}^1, \theta_j \in \mathbb{S}^1 : |\theta_{1,2}| \in \overline{\Upsilon}\}$ , meaning that, the probability that the chain revisits  $S_G^{\Psi}(\gamma_{\max})$  infinitely often is zero. First, assume that  $E[\tilde{\alpha}_{1,2}] = \mu > \sqrt{\frac{2\sigma^2}{\pi}}$ . We verify the conditions under which the chain returns to  $S_G^{\Psi}(\gamma)$  with probability one. Take  $V = |\theta_{1,2}|$  and assume that  $|\theta_{1,2}(\mathbf{k})| > \gamma$ . Calculating the drift of V based on (14), we obtain

$$\Delta V(\mathbf{k}) = \mathbf{E}[V(\mathbf{k}+1)] - V(\mathbf{k}) = \mathbf{E}[|\theta_{1,2}(\mathbf{k}+1)|] - |\theta_{1,2}(\mathbf{k})|.$$

We have  $E[|\theta_{1,2}(k+1)|] \le E[|a|] + \tau E[|b|]$ . Using the formula for the folded normal distribution [52] [see Lemma 4, inequality (28)], we obtain

$$\boldsymbol{E}[|\boldsymbol{a}|] \leq |\boldsymbol{E}[\boldsymbol{a}]| + 2\kappa\tau\sqrt{\frac{2\sigma^2}{\pi}}|\Psi(\theta_{1,2}(\mathbf{k}))|.$$

First, assume E[a] > 0. Also, without loss of generality, assume that  $\theta_{1,2}(\mathbf{k}) > 0$ , thus  $\Psi(\theta_{1,2}(\mathbf{k})) > 0$ . We obtain

$$\Delta V(\mathbf{k}) \le -2\kappa\mu\Psi(\theta_{1,2}(\mathbf{k})) + 2\kappa\sqrt{\frac{2\sigma^2}{\pi}}\Psi(\theta_{1,2}(\mathbf{k})) + \boldsymbol{E}[|b|].$$

Thus,  $\Delta V < 0$  holds if

$$\kappa > \frac{\boldsymbol{E}[|\tilde{\omega}_{1,2}(\mathbf{k})|]}{2\Psi(\theta_{1,2}(\mathbf{k}))\left(\mu - \sqrt{\frac{2\sigma^2}{\pi}}\right)}.$$
(15)

To obtain a lower bound for  $\kappa$ , upper-bound of  $E[\tilde{\omega}_{1,2}]$ and lower-bound of  $\Psi(\theta_{1,2})$  are needed in (15). Since we assumed that at time k,  $|\theta_{1,2}(k)| > \gamma$ , then either  $|\theta_{1,2}(k)| \in \Upsilon$  or  $|\theta_{1,2}(\mathbf{k})| \in \overline{\Upsilon}$ . With the former, we have  $\min \Psi(\theta_{1,2}) = |\Psi(\gamma)|$ . However, for the latter case,  $\min \Psi(\theta_{1,2}) = 0$  holds, which leads to the requirement of  $\kappa \to \infty$  to guarantee a negative drift. Now, let us set  $\min \Psi(\theta_{1,2}) = |\Psi(\gamma)|$  in (15). As discussed above, the obtained  $\kappa$  is not sufficiently large to guarantee the return of the chain from  $\overline{\Upsilon}$ , in this example equivalent to  $S_G^{\Psi}(\gamma_{\max})$ , to  $S^G(\gamma)$ . The question is whether  $S_G^{\Psi}(\gamma_{\max})$  is an absorbing set. To verify, take  $\bar{V} = \pi - |\theta_{1,2}|$  and assume that at time k,  $|\theta_{1,2}| < \gamma_{\text{max}}$ . Calculating the drift of V, we obtain  $\Delta \overline{V}(\mathbf{k}) = |\theta_{1,2}(\mathbf{k})| - E[|\theta_{1,2}(\mathbf{k}+1)|]$ . The above gives  $\Delta \overline{V}(\mathbf{k}) = -\Delta V(\mathbf{k})$ . Thus, finding a bound for  $\kappa > 0$  to guarantee  $\Delta \overline{V} < 0$  is not possible. In particular, the lower bound for  $\kappa$  in (15), with min  $\Psi(\theta_{1,2}) = |\Psi(\gamma)|$ , guarantees  $\Delta V < 0$ , and thus,  $\Delta \overline{V} > 0$ . Based on [29, Th. 8.0.2], the chain is transient on  $S_G^{\Psi}(\gamma_{\max})$ . This implies that the probability that the chain revisits  $S_G^{\Psi}(\gamma_{\max})$  infinitely often is zero. This example shows the positive effect of the additive uncertainties in the stabilization of the chain with respect to the in-phase set  $S_G^{\Psi}(\gamma)$ . Notice that for  $\boldsymbol{E}[\tilde{\alpha}_{1,2}] < 0$ , based on a similar argument, we can prove the stability of the chain with respect to the antiphase set  $S_G^{\Psi}(\gamma_{\max})$ .

We are now ready to state our main result.

*Theorem 1:* Consider the relative phase stochastic process (13) under Assumptions 1 and 2. The following statements hold.

1) If  $E[\alpha]$  is positive definite, then the relative phase process is stochastic phase-cohesive with respect to the in-phase

set  $S_G^{\Psi}(\gamma)$ , defined in (8), provided that  $\mu_m > \sqrt{\frac{2\sigma^2}{\pi}}$ ,

$$\kappa > \frac{\boldsymbol{E}_{\max}[|\Delta\tilde{\omega}|]}{\left(\mu_m - \sqrt{\frac{2\sigma^2}{\pi}}\right)|\Psi(\gamma)|\lambda_{\min}(L_e(G_{\tau}))}$$
(16a)

and  $\tau$  satisfies

$$\tau < \frac{\gamma}{\kappa \left(\mu_M + \sqrt{\frac{2\sigma^2}{\pi}}\right) \Psi_{\max} \lambda_{\max}(L_e) + \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}.$$
(16b)

If *E*[α] is negative definite and B<sup>+</sup>θ ≥ γ<sub>max</sub>1<sub>m</sub> is feasible for G(V, E), then the relative phase process is stochastic phase-cohesive with respect to the antiphase set S<sup>Ψ</sup><sub>G</sub>(γ<sub>max</sub>), defined in (9), provided that

$$\kappa > \frac{|\boldsymbol{E}_{\max}[\Delta \tilde{\omega}]|}{|\Psi(\gamma)|\,\mu_m\,\lambda_{\min}(L_e(G_\tau))} \tag{17a}$$

and  $\tau$  satisfies

$$\tau < \frac{(\pi - \gamma_{\max})}{\kappa \Psi_{\max} \,\mu_M \,\lambda_{\max}(L_e) + \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]} \quad (17b)$$

where  $\mu_m = \lambda_{\min}(|\boldsymbol{E}[\alpha]|), \ \mu_M = \lambda_{\max}(|\boldsymbol{E}[\alpha]|), \ \boldsymbol{E}_{\max}[\cdot] = \max_{i,j} \boldsymbol{E}[\cdot]; \ \lambda_{\max}(L_e)$  is the largest eigenvalue of  $L_e(G)$ ; and  $\lambda_{\min}(L_e(G_{\tau}))$  is the minimum among the smallest eigenvalues of all spanning trees of G.

Sketch of the Proof: The proof is based on an application of [51, Th. 9.1.8] in a network setting. The key is the construction of a positive and radially unbounded function  $V : \Pi \to \mathbb{R}^+$  for the discrete-time network, such that the one-step drift of V is negative if sufficient coupling conditions are satisfied. The detailed proof is provided in Appendix D.

Theorem 1 presents the coupling conditions under which the relative phase process is stochastic phase-cohesive. The main condition is the lower bound on  $\kappa$ . The bound on  $\tau$  ensures that the sampling time is sufficiently small such that given a sufficiently large  $\kappa$ , the expectation of the maximum relative phase at each time-step is confined within the desired arc. We now continue by characterizing the coupling conditions under which ultimate phase-cohesiveness is achieved. We show that for the ultimate case, the coupling condition depends on  $\tau$ . Without loss of generality, the following result is presented for the case of  $E[\alpha] > 0$ .

Corollary 1 (Ultimate stochastic phase-cohesiveness): Consider the discrete-time Markov chain in (13) representing the relative phase dynamics of n interconnected oscillators under Assumptions 1 and 2. Assume that  $E[\alpha] > 0$  holds. Then, the relative phase process is ultimate stochastic phase-cohesive with respect to the in-phase set  $S_G^{\Psi}(\gamma)$ , in (8), if the following conditions hold:

$$\kappa > \frac{\frac{1}{\tau |\Psi(\gamma)|} + \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}{\left(\mu_m - \sqrt{\frac{2\sigma^2}{\pi}}\right) |\Psi(\gamma)| \lambda_{\min}(L_e(G_{\tau}))}$$
(18a)  
$$\tau < \frac{\gamma - \frac{1}{m\Psi_{\max}}}{\kappa \left(\mu_M + \sqrt{\frac{2\sigma^2}{\pi}}\right) \Psi_{\max} \lambda_{\max}(L_e) + \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}$$
(18b)

where  $\mu_m = \lambda_{\min}(|\boldsymbol{E}[\alpha]|), \ \mu_M = \lambda_{\max}(|\boldsymbol{E}[\alpha]|), \ \boldsymbol{E}_{\max}[\cdot] = \max_{i,j} \boldsymbol{E}[\cdot]; \ \lambda_{\max}(L_e)$  is the largest eigenvalue of  $L_e(G)$ ; and  $\lambda_{\min}(L_e(G_{\tau}))$  is the minimum among the smallest eigenvalues of all spanning trees of G.

#### *C. Mixed Positive and Negative Multiplicative Mean Values*

Theorem 1 has proved that the relative phase Markov chain is stochastic phase-cohesive with respect to either the in-phase set  $S^{\Psi}_G(\gamma)$  or antiphase set  $S^{\Psi}_G(\gamma_{\max})$  depending on the sign of the mean values of the multiplicative uncertainties. The presence of mixed positive and negative mean values for the uncertain couplings could lead to a positive or negative drift condition. In this section, we study the stochastic stability of a network with an underlying connected and undirected line topology, i.e., a subclass of connected graphs without cycles in which each oscillator is connected to maximum two other oscillators. We obtain conditions under which all relative phases with positive multiplicative mean-values are recurrent to the in-phase set and those with negative mean-values are recurrent to the antiphase set. We consider identical exogenous frequencies and assume zero mean value for the additive uncertainties, i.e.,  $\boldsymbol{E}[\tilde{\omega}_{i,j}(\mathbf{k})] = 0 \forall (i,j).$  Recall the arcs  $\underline{\Upsilon}$  and  $\overline{\Upsilon}$ , defined in (5) and (7), respectively. Define,

$$U_{G}^{\Psi}(\gamma) = \{\theta_{i} \in \mathbb{S}^{1}, \theta_{j} \in \mathbb{S}^{1} \colon |\theta_{i,j}(\mathbf{k})| \in \underline{\Upsilon} \cup \overline{\Upsilon} \forall (i,j) \in \mathcal{E}\}.$$
(19)

Proposition 1: Consider the discrete-time Markov chain in (13) representing the relative phase dynamics of n interconnected oscillators over a connected and undirected line network under Assumptions 1 and 2. Assume that  $E[\tilde{\omega}_{i,j}(\mathbf{k})] = 0 \forall (i, j)$ , and each of the multiplicative uncertainties obeys  $\tilde{\alpha}_{i,j} \sim \mathcal{N}(\mu_{i,j}, \sigma^2)$ , where  $\mu_{i,j}$  is either positive or negative such that  $|\mu_{i,j}| = \lambda > 0$ , and  $\lambda \gg \sigma^2$ . If

$$\kappa > 0 \quad {
m and} \quad \tau < rac{\gamma}{2\kappa\lambda\Psi_{
m max}}$$

holds, then the relative phase process is stochastic phasecohesive with respect to  $U_G^{\Psi}(\gamma)$  in (19) such that each relative phase whose uncertain coupling weight has a positive (negative) mean value is recurrent to the in-phase arc  $\underline{\Upsilon}$  in (5) [antiphase arc  $\Upsilon$  in (7)].

#### D. Relaxation of the Odd Coupling Function

Definition of  $\underline{\Upsilon}$  in (5) assumes that  $0 \in \underline{\Upsilon}$  and  $\Psi(0) = 0$ . However, the latter condition can be relaxed such that the coupling function takes a zero value at a nonzero arc  $\gamma_c : 0 < \gamma_c < \pi$ . This allows considering cases where the coupling is not an odd function on the entire interval  $[-\pi, \pi]$ .

Assumption 1' Function  $\Psi_r(\cdot)$  is  $2\pi$ -periodic, and continuously differentiable. There exists an arc  $\Upsilon \subset (0, \pi)$  such that  $\forall \xi, z \in [0, \pi]$ , if  $\xi \notin \Upsilon$ ,  $z \in \Upsilon$ , then  $|\Psi_r(\xi)| \leq |\Psi_r(z)|$ . In the interval  $(0, \pi)$ , there exists  $\gamma_c \notin \Upsilon$  such that  $\gamma_c < \gamma \forall \gamma \in \Upsilon$ , and it holds that  $\gamma_c$  and  $\pi$  are the only roots of  $\Psi_r(\cdot)$  in  $[\gamma_c, \pi]$ . Moreover, function  $\Psi_r(\cdot)$  is odd on  $[-\pi, -\gamma_c] \cup [\gamma_c, \pi]$ , and  $\forall \gamma_i \in [-\gamma_c, \gamma_c] : |\Psi_r(\gamma_i)| \leq \overline{\Psi} < |\Psi_r(\gamma)|$ .

Following Assumption 1', we construct the two following sets:

$$\underline{\Upsilon}^1 = [0, \gamma_c] : \{ \forall \gamma_i \in \underline{\Upsilon}^1 : 0 \le |\Psi_r(\gamma_i)| \le \bar{\Psi} \}$$
(20)

$$\underline{\Upsilon}^2 = [\gamma_c, \gamma] : \{ \forall \gamma_i \in \underline{\Upsilon}^2 : 0 \le |\Psi_r(\gamma_i)| \le |\Psi_r(\gamma)| \}.$$
(21)

Accordingly, we modify the definition of  $S_G^{\Psi}(\gamma)$  as follows:

$$S_{G}^{\Psi_{r}}(\gamma) = \{\theta_{i}, \theta_{j} \in \mathbb{S}^{1} : |\theta_{i,j}(\mathbf{k})| \in \underline{\Upsilon}^{1} \cup \underline{\Upsilon}^{2} \forall (i,j) \in \mathcal{E}\}.$$
(22)

Proposition 2: Consider the discrete-time Markov chain in (13) over a tree network topology and under Assumptions 1' and 2. Assume that  $E[\alpha]$  is positive definite, and  $\lambda_{\min}(E[\alpha]) =$  $\mu_m \gg \sigma^2$  holds. If  $\hat{\lambda} = \mu_m |\Psi_r(\gamma)| - \mu_M \bar{\Psi} \sqrt{(m-1)} > 0$ , then the relative phase process is stochastic phase-cohesive, with the absorbing set  $S_G^{\Psi_r}(\gamma)$  in (22) provided that

$$\kappa > \frac{\left(\hat{\lambda}_1 + \hat{\lambda}_2(m-1)\right) \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}{\hat{\lambda} \lambda_{\min}(L_e) \left(\hat{\lambda}_1 + \hat{\lambda}_2 \sqrt{(m-1)}\right)}$$
(23a)

$$\tau < \frac{\gamma}{\kappa \Psi_{\max} \lambda_{\max}(L_e) \ \mu_M + \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}$$
(23b)

where  $\mu_m = \lambda_{\min}(\boldsymbol{E}[\alpha]), \quad \mu_M = \lambda_{\max}(\boldsymbol{E}[\alpha]), \quad \boldsymbol{E}_{\max}[\cdot] =$  $\max_{i,i} \boldsymbol{E}[\cdot], \ \hat{\lambda}_1 = \mu_m |\Psi_r(\gamma)|, \ \hat{\lambda}_2 = \mu_M \bar{\Psi}, \ \text{and} \ \lambda_{\max}(L_e) \ \text{and}$  $\lambda_{\min}(L_e)$  are, respectively, the largest and smallest eigenvalues of the underlying tree graph's edge Laplacian.

*Remark 1:* [Insights from stochastic stability analysis] Problem 1 considers stochastic uncertainties, with continuous probability distributions, which can take positive, negative, or zero values at every sample time with no restriction on their amplitudes. Compared with the deterministic stability, which depends on either the upper bound [43] (for additive), or the bound, sign, and behavior (for multiplicative) of the time-evolution of disturbances [44], our results only require finite means and bounded variances. To elaborate further, consider phase oscillators in a network where the mean-values of multiplicative uncertainties are positive, and the mean values of additive noises are zero, i.e.,  $E[\alpha] > 0$  and  $E_{\max}[|\Delta \tilde{\omega}|] = 0$ . Based on Theorem 1, for any  $\kappa > 0$ , the in-phase set is stochastic phase-cohesive. This also includes the case of  $\gamma = 0$ , indicating stability of the phase-locked (exact synchronization) solution, despite the fact that the samples of all uncertainties can take any value at any time step. In comparison, for the deterministic Kuramoto model with the all-to-all topology, local input-to-state stability of the phase-locked solution given bounded additive perturbations, has been proved [43]. We also notice that despite the effects of multiplicative uncertainties on the network topology at each sample time, our stochastic analysis does not require considering various possible time-varying typologies. Next section studies a specific model of multiplicative uncertainties, i.e., random connections. We show that stochastic phase-cohesiveness depends on the probability of interconnections of oscillators, comparable with results to the stochastic linear consensus problem [35], and different from the deterministic setting, where stability depends on the time-varying graphs.

#### V. OSCILLATORS IN A RANDOM NETWORK

In this section, we investigate Problem 2 in order to characterize conditions under which the stochastic phase-cohesiveness is achieved for the coupled oscillators over an Erdös-Rényi random network. Considering each oscillator's dynamics in (11), the relative phase dynamics of oscillators i and j obeys

a (\* )

o (\*

$$\theta_{i,j}(\mathbf{k}+1) = \theta_{i,j}(\mathbf{k}) + \tau \omega_{i,j} - \left(\kappa \tau \sum_{\ell \in \{i,j\}} \sum_{e \in \mathcal{N}_{\ell}} \beta_{\ell,e}(\mathbf{k}) \Psi(\theta_{\ell,e}(\mathbf{k}))\right)$$
(24)

where  $\theta_{i,j}(\mathbf{k})$  and  $\omega_{i,j} \in \mathbb{R}$  represent the relative phase and relative exogenous frequency of the two oscillators i and j, respectively. The random variable  $\beta_{\ell,e}(\mathbf{k})$  obeys the Bernoulli distribution, i.e.,

$$\beta_{\ell,e}(\mathbf{k}) = \begin{cases} 1 & \text{with probability } p \\ 0 & \text{with probability } 1 - p. \end{cases}$$
(25)

Considering a probable absence of each edge in the network, a graph with the incidence matrix  $B^{k}$  presents the topology of the network at time k. We denote the set of incidence matrices associated with random graphs, with n nodes and probability of link failure of 1 - p, by  $\mathbf{B}(n, p)$ . Thus,  $B^{k} \in \mathbf{B}(n, p)$ . To stay consistent with the problem formulation of the previous section and without loss of generality, we assume the existence of a maximal graph for the given problem, for instance a complete graph, denoted by G whose incidence matrix is B. We then model the randomness with the term  $B\beta(\mathbf{k})$ , where  $\beta_{m \times m}(\mathbf{k})$ is a diagonal matrix capturing the random interconnections.

Thus, the relative phase vector for the whole network,  $\Theta(k)$  in (3), is equal to  $B^{+}\theta(\mathbf{k})$ . The compact form of the relative phase dynamics follows

$$B^{\top}\boldsymbol{\theta}(\mathbf{k}+1) = B^{\top} \left(\boldsymbol{\theta}(\mathbf{k}) + \tau \boldsymbol{\omega}(\mathbf{k}) - \tau \kappa B \boldsymbol{\beta}(\mathbf{k}) \Psi(B^{\top}\boldsymbol{\theta}(\mathbf{k}))\right)$$
(26)

where  $\beta(k)$  is a diagonal matrix whose diagonal elements obey (25) and

$$\boldsymbol{\omega}_{n \times 1}(\mathbf{k}) = (\omega_1(\mathbf{k}), \dots, \omega_n(\mathbf{k}))^\top$$

The above model allows us to study the stochastic phasecohesiveness of the network by using the developed setting in the previous sections. Different from Section IV, this section focuses on the case of constant exogenous frequencies. We now discuss the chain properties and show that  $S_G^{\Psi}(\gamma)$  in (5) is the stable set for this model as well.

Lemma 3: The relative phase stochastic process in (24) is a  $\psi$ -irreducible Markov chain on a countable space.

The following result presents a counterpart of Theorem 1 for the case of random networks.

Theorem 2: Consider the discrete-time Markov chain in (26) under Assumption 1. Assume that  $\omega_{i,j} \neq 0 \forall i, j$ . If p is strictly positive, then the relative phase process is stochastic phasecohesive with respect to  $S_G^{\Psi}(\gamma)$  in (8) if the following conditions hold

$$\kappa > \frac{|\Delta_{\max}\omega|}{|\Psi(\gamma)| \ p \ \lambda_{\min}(L_e(G_\tau))}$$
(27a)

$$\tau < \frac{\gamma}{\kappa \,\Psi_{\max} \,\lambda_{\max}(L_e)} \tag{27b}$$

where  $|\Delta_{\max}\omega| = \max_{i=1} |\omega_i - \omega_j|$ ,  $\lambda_{\max}(L_e)$  is the largest eigenvalue of L(G), and  $\lambda_{\min}(L_e(G_{\tau}))$  is the minimum among the smallest eigenvalues of the spanning trees of G. 

We now assume identical exogenous frequencies and derive conditions under which the chain achieves phasesynchronization (or phase-locking). With this, we show the applicability of our definition of stochastic phase-cohesiveness for a stronger notion of synchronization, i.e., phase-locking.

*Corollary 2:* Consider the discrete-time Markov chain in (26) under Assumption 1 and with identical exogenous frequencies. Assuming a sufficiently small  $\tau$ , the relative phase Markov chain



Fig. 2. Plot of  $\Psi(\theta)$  (left) and  $\Psi_r(\theta)$  (right). The "x" axis ranges over  $[-\pi, \pi]$ . The red dots on  $\Psi(\theta)$  show the minimum and maximum possible values for  $\gamma$ . The red circle on  $\Psi_r(\theta)$  shows where the interval in which function is not odd, and the red dot shows  $\gamma_c$ .



Fig. 3. Network of five oscillators subject to multiplicative and additive uncertainties. Dashed arrows illustrate assignment of exogenous frequencies.

is stochastic phase-cohesive with respect to the origin provided that p > 0 and  $\kappa > 0$ .

*Remark 2:* It is worth noting the difference between the probability spaces for the uncertain network (Problem 1) and the random network (Problem 2). In Problem 1, the relative phases at time k evolve in a general space which is generated by independent stochastic processes governing their corresponding exogenous frequencies and couplings. In Problem 2, however, the randomness only affects the interconnection topology. As a result, the relative phases at time k will transit to a new state within a countable set of states generated by independent Bernoulli processes determining the interconnection topology.

#### **VI. SIMULATION RESULTS**

This section presents numerical simulations to validate our theoretical results on the stochastic phase-cohesive behavior of interconnected oscillators for both uncertain (Section IV) and random (Section V) networks. We assume that the coupling law obeys  $\Psi(\theta) = \sin(\theta) + 0.3 \sin(3\theta)$  which meets Assumption 1. We also define the arcs  $\underline{\Upsilon} = [0, \frac{\pi}{8}], \overline{\Upsilon} = [\frac{\pi}{1.14}, \pi]$ . Fig. 2 shows the plot of function  $\Psi(\theta)$  as well as  $\Psi_r(\theta)$ . The latter,  $\Psi_r(\theta) = 1.5 \sin(1.1\theta) - 0.7 \cos(3.3\theta - 0.4\pi)$ , is not an odd function on the entire interval  $[-\pi, \pi]$ . This function is used in simulations designed for the verification of Proposition 2.

#### A. Uncertain Network

Fig. 3 shows a network composed of five oscillators. The constant components of the exogenous frequencies are set to  $\omega_1 = 1, \omega_2 = 2, \omega_3 = 3, \omega_4 = 4, \omega_5 = 5$ . The initial conditions



Fig. 4. Phases, relative phases and the maximum relative phase: uncertain couplings (positive mean), and exogenous frequencies (Theorem 1.1).

for the oscillators are set to  $\theta(0) = [\frac{\pi}{4}, \frac{\pi}{8}, \frac{-\pi}{8}, \frac{-\pi}{5}, \frac{\pi}{5}]$ . The multiplicative and additive stochastic uncertainties are modeled by the Gaussian random variables and reported by  $\tilde{\alpha}_{\ell,e}^{(m,v)} =$  (Mean value, Variance) and  $\varpi_i^{(m,v)} =$  (Mean value, Variance). For the first experiment, the mean values and variances are set based on the following table. Calculating the bound for  $\kappa$  based on Theorem 1, we obtain  $\kappa > 39.8$  by replacing  $E_{\max}[|\Delta\tilde{\omega}|]=3$ ,  $\mu_m - \sqrt{\frac{2\sigma^2}{\pi}}s=0.3$ ,  $|\Psi(\gamma)|=0.66$ , and  $\lambda_{\min}(L_e(G_{\tau}))=0.38$ . The latter is the minimum eigenvalue of the network's spanning tree, which is a line graph obtained by removing the edge (3,4). We set  $\kappa = 40$  and  $\tau = 0.001$  meeting the requirements of Theorem 1.

$\widetilde{\alpha}_{\ell,e}^{(m,v)}$	$arpi_i^{(m,v)}$
$\tilde{\alpha}_{1,2}^{(m,v)} = (1,0.5)$	$\varpi_1^{(m,v)} = (4,1)$
$\tilde{\alpha}_{2,3}^{(m,v)} = (3,0.5)$	$\varpi_2^{(m,v)} = (2,2)$
$\tilde{\alpha}_{3,4}^{(m,v)} = (0.85, 0.5)$	$\varpi_3^{(m,v)} = (0,1)$
$\tilde{\alpha}_{1,4}^{(m,v)} = (1.5, 0.5)$	$ \varpi_4^{(m,v)} = (1,3) $
$\tilde{\alpha}_{4,5}^{(m,v)} = (2,0.5)$	$ \varpi_5^{(m,v)} = (-2, 1.5) $

The time-evolution of the oscillators' phases, the relative phases, and the maximum relative phase are shown in Fig. 4. As shown the relative phases are confined in the desired set.

In order to examine the effects of multiplicative uncertainties with negative mean values, we keep the settings of the first experiment but replace the mean values of the multiplicative randomness with negative values. The time evolution of relative phases, where all multiplicative mean values are set to negative ones are shown in Fig. 5. The results confirm the stochastic phase-cohesiveness w.r.t the antiphase set.

To verify Propositions 1 and 2, we consider a line graph obtained by removing the edge (3,4) of the graph shown in Fig. 3. We examine the results of Proposition 1 by setting the size of all mean values of the multiplicative uncertainties equal to one. We set  $\tilde{\alpha}_{1,2}^{(m,v)} = (-1,0.5)$ ;  $\tilde{\alpha}_{2,3}^{(m,v)} = (-1,0.5)$  and keep the rest of the mean values positive. We also set  $\boldsymbol{E}[\tilde{\omega}_i] = 0 \forall i$ . The coupling coefficient  $\kappa > 0$  is set to  $\kappa = 2$ . As shown in Fig. 6, the relative phases form two clusters.

To verify the result of Proposition 2, we use the setting of the first experiment but with a nonodd coupling function  $\Psi_r(\theta)$ . We set  $\Psi_r(\gamma = 0.4\pi) = 2$  and  $\overline{\Psi} = 0.2$ . We have  $\mu_m = 1$ ;  $\mu_M = 3$ ; m = 4, which gives  $\hat{\lambda} = 0.98$  and  $\kappa > 10$ . We set  $\kappa = 10$ . As



Fig. 5. Phases, relative phases and the maximum relative phase: couplings (negative mean values), and uncertain exogenous frequencies (Theorem 1.2).



Fig. 6. Phases and relative phases over a line network with multiplicative uncertainties with mixed positive and negative mean values (Proposition 1).



Fig. 7. Phases, relative phases, the maximum relative phase: Coupled by  $\Psi_r(\theta)$ , uncertain exogenous frequencies, and couplings with positive mean values (Proposition 2).

shown in Fig. 7, the relative phases are bounded and the network behavior follows the result of Proposition 2.

#### B. Random Network

We now present the simulation results for a random network with five oscillators. We use the coupling function  $\Psi(\cdot)$ ,  $\underline{\Upsilon} = [0, \frac{\pi}{8}]$ , and the initial conditions of the oscillators similar to the first experiment of the previous section. If all links are connected, then the graph depicted in Fig. 3 is obtained, hence the maximal graph. The nonzero and nonidentical constant exogenous frequencies are equal to  $\omega_1 = 1, \omega_2 = 2, \omega_3 = 3, \omega_4 =$ 4, and  $\omega_5 = 5$ . Sampling time is set to 0.01 s to elaborate the effects of the randomness. We first assume that p = 0.8. Using the condition in (27), we calculate  $\kappa > 15$  and set  $\kappa = 19$ . The evolution of the oscillators' phases, the relative phases, and the maximum relative phase over time are shown in Fig. 8(a). As shown, the relative phases are confined in the set  $S_{\underline{W}}^{\Psi}(\gamma = \frac{\pi}{8})$ .

We then decrease the connectivity probability to 0.3. The results, reported in Fig. 8(b), show that the relative phases are not bounded. We update the value of  $\kappa$  with respect to the decrease in the connection probability to  $\kappa = 30$ . The plots in Fig. 8(c) show that the maximum relative phase is now within the desired set.



Fig. 8. Phases, relative phases and the maximum relative phase over random network: nonidentical exogenous frequencies: (a) p = 0.8,  $\kappa = 19$ , (b) p = 0.3,  $\kappa = 12$ , (c) p = 0.3,  $\kappa = 30$  (Theorem 2).



Fig. 9. Phases, relative phases and the maximum relative phase over random network: identical exogenous frequencies,  $\kappa = 0.5$ : (a) p = 0.8, (b) p = 0.1.

Next, we assume all exogenous frequencies are set to one. We set  $\kappa = 0.5 > 0$ , and consider two probabilities of connection: p = 0.8 and p = 0.1. Plots in Fig. 9 show the time-evolution of the phases, the relative phases, and the maximum relative phase over time for these two cases. As shown, all oscillators' relative phases converge to zero and the rate of convergence is proportional to the probability of connection.

#### VII. CONCLUSION

This article studies the stochastic relative phase stability for a class of discrete-time coupled oscillators. The two notions of stochastic phase-cohesiveness and ultimate stochastic phasecohesiveness are introduced. Stochastic phase-cohesiveness of oscillators, with a general class of  $2\pi$ -periodic, and odd coupling functions, with respect to the two, in-phase and antiphase, sets were studied. We investigated undirected networks subject to both multiplicative and additive stochastic uncertainties. We proved stochastic phase-cohesiveness with respect to the inphase set when the mean values of all multiplicative uncertainties were positive, and with respect to the antiphase set for the case of negative mean values. In addition, we have discussed the relaxation of the odd property of the coupling function by allowing this function to be nonodd on a subset of its domain. Moreover, we proved a clustering behavior for a network with an underlying line topology subject to mixed negative and positive mean values for the multiplicative uncertainties and zero mean value for the additive uncertainties. Further, the stochastic phasecohesiveness of oscillators with constant exogenous frequencies in an Erdös-Rényi random network was studied. Sufficient conditions for achieving both stochastic phase-cohesive and phase-locked solutions were derived. It was proved that oscillators with equal exogenous frequencies in a random network with any positive possibility of connection will achieve phaselocking. Our results emphasize the importance of the coupling function in synchronization, discuss the stabilizing effects of the additive stochastic uncertainties, and the effects of multiplicative uncertainties in achieving stochastic phase-cohesiveness.

# APPENDIX A PROOF OF LEMMA 1

*Proof:* The relative phase process  $\Theta(\mathbf{k}) = B^{\top} \boldsymbol{\theta}(\mathbf{k})$  with i.i.d. stochastic variables, generated by continuous distributions, satisfies all properties in Lemma 1 according to the definition of the Markov chains, and Definitions 2 and 3.

# APPENDIX B PROOF OF LEMMA 2

**Proof:** Based on [29, Proposition 6.2.8], the satisfaction of the Feller property (Definition 5) together with the nonemptiness of the support of the irreducibility measure leads to the conclusion that every compact set in the state space is also petite. The Feller property of the Markov chain (13) can be readily concluded according to [29, Proposition 6.1.2], since the right-side of the Markov chain (13) is a continuous function in  $\theta(k)$  for each fixed pair of i.i.d. realizations ( $\tilde{\omega}(k), \alpha(k)$ ). In addition, the uncertainties are capable of forcing transitions from any subset of the  $\sigma$ -algebra to an open petite set in the state space, hence, there exists no set in the  $\sigma$ -algebra in which if the Markov state enters, it always remains there with the absolute probability of one. This concludes that the support of the irreducibility measure has a nonempty interior, which completes the proof.  $\Box$ 

#### APPENDIX C LEMMA 3: STATEMENT AND PROOF

Lemma 4: Consider the random vector

$$y(\mathbf{k}) = B^{\top} \boldsymbol{\theta}(\mathbf{k}) - \tau \kappa B^{\top} B \boldsymbol{\alpha}(\mathbf{k}) \Psi(B^{\top} \boldsymbol{\theta}(\mathbf{k}))$$

where  $y(\mathbf{k}) \in \mathbb{R}^{m \times 1}$  and  $\alpha(\mathbf{k})$  is a diagonal matrix whose elements are i.i.d Gaussian random variables, as in Assumption 2. Then, the following inequality holds:

$$\begin{split} \boldsymbol{E}[|\boldsymbol{y}(\mathbf{k})| | \boldsymbol{\theta}(\mathbf{k})] &\leq |\boldsymbol{E}[\boldsymbol{y}(\mathbf{k}) | \boldsymbol{\theta}(\mathbf{k})]| \\ &+ \kappa \tau \sqrt{\frac{2\sigma^2}{\pi}} |\boldsymbol{B}^\top \boldsymbol{B} \boldsymbol{\Psi}(\boldsymbol{B}^\top \boldsymbol{\theta}(\mathbf{k}))|. \end{split}$$

*Proof:* The proof is based on an application of the folded normal distribution [52], for calculation of the expectation of the absolute value of a random variable  $Z \sim N(\mu, \sigma^2)$ . We have

$$\boldsymbol{E}[|Z|] = \sqrt{\frac{2\sigma^2}{\pi}} \exp\left(\frac{-\mu^2}{2\sigma^2}\right) + \mu \operatorname{erf}\left(\frac{\mu}{\sqrt{2\sigma^2}}\right)$$

where erf, the error function [52], is an odd function such that  $|erf(.)| \le 1$ . As a result, we can write

$$\boldsymbol{E}[|\boldsymbol{Z}|] \le \sqrt{\frac{2\sigma^2}{\pi}} + |\boldsymbol{\mu}|. \tag{28}$$

Now, each element of  $y(\mathbf{k}) = [y_1(\mathbf{k}) \dots y_m(\mathbf{k})]^\top$  is a random variable obeying the normal distribution. Calculating  $E[|y(\mathbf{k})||\boldsymbol{\theta}(\mathbf{k})]$ , requires computation of the expectation of absolute value of each element of  $y(\mathbf{k})$ , i.e.,  $E[|y_i(\mathbf{k})||\boldsymbol{\theta}(\mathbf{k})]$ . Denote  $B^\top B\Psi(B^\top\boldsymbol{\theta}(\mathbf{k}))$  by  $x = [x_1 \dots x_m]^\top$ . We have

$$\boldsymbol{E}[|y_i(\mathbf{k})| | \boldsymbol{\theta}(\mathbf{k})] \le |\boldsymbol{E}[y_i(\mathbf{k})| \boldsymbol{\theta}(\mathbf{k})]| + \sqrt{\frac{2\sigma^2 \kappa^2 \tau^2 x_i^2}{\pi}}$$

The proof is completed by stacking all  $y_i$  into vector y(k), and all  $x_i$  into x.

### APPENDIX D PROOF OF THEOREM 1

*Proof:* The proof is based on an application of [51, Th. 9.1.8] in a network setting. We first assume that  $E[\alpha]$  is positive definite and prove that the Markov chain in (13) is stochastic phase-cohesive with respect to the in-phase set  $S_G^{\Psi}(\gamma)$ . Let us assume that the chain lives in the set

$$U^{1} = \{\theta_{i}, \theta_{j} \in \mathbb{S}^{1} : |\theta_{i,j}(\mathbf{k})| \in \underline{\Upsilon} \cup \Upsilon \forall (i,j) \in \mathcal{E}\}$$

where  $\underline{\Upsilon}$  and  $\underline{\Upsilon}$  are defined in (5) and (6), respectively.

Let us first assume that  $E[\alpha]$  is positive definite and define the positive and radially unbounded function

$$V(\Theta(\mathbf{k})) = |\Psi(\gamma)| \sum_{|\theta_{i,j}(\mathbf{k})| \in \Upsilon} |\theta_{i,j}(\mathbf{k})| + \Psi^o \sum_{|\theta_{i,j}(\mathbf{k})| \in \underline{\Upsilon}} |\theta_{i,j}(\mathbf{k})|$$
(29)

where  $\Theta(\mathbf{k}) = B^{\top} \boldsymbol{\theta}(\mathbf{k}), \Psi^{o} = \Psi(0^{+})$ , and  $V : \Pi \to \mathbb{R}^{+}$ , with m the total number of edges of the underlying graph. Notice that in the view of Lemma 2, the level sets of V are petite. Let us assume that  $\{\exists(\ell, q) \text{ s.t. } |\boldsymbol{\theta}_{\ell}(\mathbf{k}) - \boldsymbol{\theta}_{q}(\mathbf{k})| \in \Upsilon\}$ . We now derive conditions under which the one-step drift of V, i.e.,

$$\Delta V(\Theta) = \boldsymbol{E}[V(\Theta(\mathbf{k}+1))|\Theta(\mathbf{k})] - V(\Theta(\mathbf{k}))$$
(30)

is negative. Notice that the oscillators' phases at time k, i.e.,  $\theta(k)$  is known. Let us denote the coefficients of  $V(\Theta(k))$  by  $C \in \mathbb{R}^{m \times 1}$ . Then, we can write  $V(\Theta(k)) = C^{\top} | B^{\top} \theta(k) |$ . From the relative phase dynamics in (13), we have

$$E[V(\Theta(\mathbf{k}+1))|\Theta(\mathbf{k})] = C^{\top} E[|B^{\top} \boldsymbol{\theta}(\mathbf{k}+1)||\boldsymbol{\theta}(\mathbf{k})]$$
  
=  $C^{\top} E[|B^{\top} \boldsymbol{\theta}(\mathbf{k}) - \tau \kappa B^{\top} B \underline{\alpha}(\mathbf{k}) \Psi(B^{\top} \boldsymbol{\theta}(\mathbf{k})) + \tau B^{\top} \tilde{\boldsymbol{\omega}}(\mathbf{k})|]$   
$$\leq C^{\top} \left( E[|\underbrace{B^{\top} \boldsymbol{\theta}(\mathbf{k}) - \tau \kappa B^{\top} B \underline{\alpha}(\mathbf{k}) \Psi(B^{\top} \boldsymbol{\theta}(\mathbf{k}))}_{I_{1}}| + \underbrace{\tau | B^{\top} \tilde{\boldsymbol{\omega}}(\mathbf{k})|}_{I_{2}}\right)$$
(31)

where  $I_1 = (I_1^1, \dots, I_1^m)^\top \in \mathbb{R}^{1 \times m}$  and  $I_2 \in \mathbb{R}^{m \times 1}$ . According to Lemma 4, we have

$$\boldsymbol{E}[|I_1||\boldsymbol{\theta}(\mathbf{k})] \leq |\boldsymbol{E}[I_1|\boldsymbol{\theta}(\mathbf{k})]| + \kappa\tau \sqrt{\frac{2\sigma^2}{\pi}} |B^{\top}B\Psi(B^{\top}\boldsymbol{\theta}(\mathbf{k}))|.$$
(32)

For the clarity of presentation, we denote  $V(\Theta(\mathbf{k}))$  and  $\Psi^{\top}(B^{\top}\boldsymbol{\theta}(\mathbf{k}))$  by  $V(\mathbf{k})$  and  $\Psi^{\top}(\mathbf{k})$ , respectively. Also, define  $r = \kappa \tau \sqrt{\frac{2\sigma^2}{\pi}}$ . Therefore from (30), we can write

$$\Delta V(\Theta) \le C^{\top} \left( |\boldsymbol{E}[I_1]| + r |\boldsymbol{B}^{\top} \boldsymbol{B} \boldsymbol{\Psi}^{\top}(\mathbf{k})| + \boldsymbol{E}[I_2] \right) - V(\mathbf{k}).$$
(33)

Based on the definition of  $\Psi^o$ ,  $\Psi^o \leq |\Psi(\theta_{i,j}(\mathbf{k}))|$  for edges with a nonzero  $\Psi(\cdot)$  at each time k. Also, since  $\Psi(\cdot)$  is an odd function, and  $\Psi(0) = 0$ , if  $\theta_{i,j}(\mathbf{k}) = 0$ , we can replace  $\Psi^o$  with  $\Psi(\theta_{i,j}(\mathbf{k})) = 0$ . Thus,  $C \leq |\Psi(B^{\top}\boldsymbol{\theta}(\mathbf{k}))|$  holds elementwise, and we can write

$$\Delta V(\Theta) \leq |\Psi^{\top}(\mathbf{k})| \left( |\boldsymbol{E}[I_1]| + r|B^{\top}B\Psi^{\top}(\mathbf{k})| + \boldsymbol{E}[I_2] \right) - V(\mathbf{k}).$$
(34)

Considering the definition of  $V(\Theta(\mathbf{k}))$  in (29), at each time k it holds that  $V_m(\Theta(\mathbf{k})) \leq V(\Theta(\mathbf{k})) \leq V_M(\Theta(\mathbf{k}))$ , where  $V_m$ represents the case in which at time k only one relative phase belongs to  $\Upsilon$ , and all other relative phases are equal to zero, and  $V_M$  is the case where all relative phases at time k belong to  $\Upsilon$ . In what follows, we continue the proof by obtaining the conditions under which both  $V_m$  and  $V_M$  decrease at one time-step. Thus, V necessarily decreases.

Proof of  $\Delta V_m(\Theta) \leq 0$ : Given the conditions of this case, only one relative phase at time k belongs to  $\Upsilon$ , and all other relative phases are equal to zero, i.e.,  $|(\theta_{\ell,q}(\mathbf{k}))| \geq \gamma$  and  $|\theta_{p,s}(\mathbf{k})| = 0 \forall (p,s) \neq (\ell,q)$ . From (34), all elements of the vector  $|\Psi^{\top}(\mathbf{k})||\mathbf{E}[I_1]|$  are zero except the element corresponding to  $\theta_{\ell,q}$ . Therefore, the following equality holds:

$$|\Psi^{\top}(\mathbf{k})||\boldsymbol{E}[I_1]| = |\Psi^{\top}(\mathbf{k})\boldsymbol{E}[I_1]|.$$

From (31) and (34), the on-step drift of  $V_m$  obeys

$$\Delta V_{m}(\Theta) \leq |\underline{\Psi^{\top}(\mathbf{k})B^{\top}\boldsymbol{\theta}(\mathbf{k})}_{a} - \underbrace{\tau\kappa \ \Psi^{\top}(\mathbf{k})B^{\top}B \ \boldsymbol{E}[\boldsymbol{\alpha}(\mathbf{k})]\Psi(\mathbf{k})}_{b}| + E[\underbrace{\tau|\Psi^{\top}(\mathbf{k})||B^{\top}\boldsymbol{\tilde{\omega}}(\mathbf{k})|]}_{c} + \underbrace{r|\Psi^{\top}(\mathbf{k})||B^{\top}B\Psi^{\top}(\mathbf{k})|}_{d} - V_{m}(\mathbf{k}).$$
(35)

To have  $\Delta V_m < 0$ , the following should hold:

[1] 
$$E[a-b+c] + d < V_m(k)$$
 when  $E[a-b] > 0$   
[2]  $E[-a+b+c] + d < V_m(k)$  when  $E[a-b] < 0.$  (36)

We now discuss that the expectation of the term denoted by b is positive. Assume  $E[\alpha]$  is positive definite. Since  $B^{\top}B \ge 0$  holds for a connected graph [50], and also  $\Psi(\cdot)$  is an odd function, we obtain

$$\boldsymbol{E}[b] \ge \kappa \, \tau \lambda_{\min}(\boldsymbol{E}[\boldsymbol{\alpha}]) \, \boldsymbol{\Psi}^{\top}(\mathbf{k}) \, \boldsymbol{B}^{\top} \boldsymbol{B} \, \boldsymbol{\Psi}(\mathbf{k}). \tag{37}$$

Notice that inequality (37) is greater than zero. The reason is that the range space of  $B^{\top}$  (i.e.,  $\mathcal{R}(B^{\top})$ ) and null space of B (i.e.,  $\mathcal{N}(B)$ ) are perpendicular. As a result,  $B\Psi(B^{\top}\boldsymbol{\theta}(\mathbf{k})) = \mathbf{0}$ 

holds if and only if  $\Psi(B^{\top}\theta(\mathbf{k})) = \mathbf{0}$ . Since we assumed that at time k, there is at least one edge of the graph whose corresponding relative phase belongs to  $\Upsilon$ ,  $B^{\top}\theta(\mathbf{k}) \neq 0$  holds. Thus, E[b] > 0. Now, consider the term d in (35). Since only one element of  $\Psi(\mathbf{k})$  is nonzero and  $B^{\top}B \geq 0$ , we can write  $d = r\Psi^{\top}(\mathbf{k})B^{\top}B\Psi^{\top}(\mathbf{k})$ .

Consider the first inequality in (36). Recall that  $|(\theta_{\ell,q}(\mathbf{k}))| \geq \gamma$  and  $|\theta_{p,s}(\mathbf{k})| = 0 \forall (p, s) \neq (\ell, q)$ . This gives  $a = \gamma \Psi(\gamma)$ . Calculating the term  $V_m(\mathbf{k})$  under the same condition, we obtain  $\gamma |\Psi(\gamma)|$  which cancels out with a. We now proceed to characterize conditions which guarantee  $\Delta V_m < 0$ . From (36), two following criteria should hold:

$$\min \{ E[b] \} - d > \max E[c] - \min \{ a \} + \max \{ E[b] + E[c] \} + d < \min \{ V_m(k) \}.$$
(38)

To obtain the lower bound of b, we write  $B^{\top}B = R^{\top}L_e(G_{\tau})R$ , where  $L_e(G_{\tau}) = B_{\tau}^{\top}B_{\tau} > 0$  is the corresponding edge Laplacian of a spanning tree of graph G at time k, and  $R = [I \ T]$ ([50, Th. 4.3], see Section II). Thus,

$$\boldsymbol{E}[b] \ge \tau \kappa \lambda_{\min}(\boldsymbol{E}[\boldsymbol{\alpha}]) \lambda_{\min}(L_e(\mathcal{G}_{\tau})) \boldsymbol{\Psi}^{\top}(\mathbf{k}) R^T R \boldsymbol{\Psi}(\mathbf{k}) > 0.$$
(39)

Under conditions  $|(\theta_{\ell,q}(\mathbf{k}))| = \gamma$  and  $|\theta_{p,s}(\mathbf{k})| = 0 \forall (p,s) \neq (\ell,q)$  and assuming that  $(\ell,q) \in \mathcal{G}_{\tau}$ , we have  $\Psi^{\top}(\mathbf{k})R^{T}R\Psi(\mathbf{k}) = \Psi^{2}(\gamma)$ . Let  $\mu_{m}$  denote  $\lambda_{\min}(\boldsymbol{E}[\alpha])$ . Computing the first inequality in (38), gives

$$(\tau \kappa \mu_m - r) \Psi^2(\gamma) \lambda_{\min}(L_e(G_\tau)) > \tau |\Psi(\gamma)| \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|].$$
(40)

Considering the second inequality in (38),  $\max\{b\}$  is obtained if  $\Psi(\theta_{\ell,q}(\mathbf{k})) = \Psi_{\max}$ . In this case, we have

$$\boldsymbol{E}[b] \leq \tau \kappa \lambda_{\max}(\boldsymbol{E}[\alpha]) \lambda_{\max}(L_e) \Psi^{\top}(\mathbf{k}) \Psi(\mathbf{k}).$$
(41)

Denoting  $\lambda_{\max}(\boldsymbol{E}[\boldsymbol{\alpha}])$  by  $\mu_M$ , we obtain

$$d + \boldsymbol{E}[b] \le (\tau \kappa \mu_M + r) \lambda_{\max}(L_e) \Psi_{\max}^2.$$

Also, calculating min  $\{a\}$ , where a is defined in (35), gives  $a \ge \Psi_{\max}\gamma$ . The reason is that for the edge belonging to  $\Upsilon$ , the minimum angle is  $\gamma$ , and its corresponding  $\Psi(\cdot)$  value follows same as in calculation of b, hence  $\Psi_{\max}$ . We notice that based on the definition of d in (29), for all edges that belong to  $\Upsilon$ , min $\{d\} = \gamma |\Psi(\gamma)|$ . Hence,

$$-\gamma + \underbrace{(\tau \kappa \mu_M + r) \lambda_{\max}(L_e) \Psi_{\max} + \tau \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}_{f} < \frac{|\Psi(\gamma)|}{\Psi_{\max}} \gamma$$
(42)

should hold. Since  $|\Psi(\gamma)| < \Psi_{\text{max}}$ , we can replace (42) with  $f < \gamma$ . Thus, we obtain the second condition in (16).

Proof of  $\Delta V_M(\Theta) \leq 0$ : Given the conditions of this case, all relative phases at time k belong to  $\Upsilon$ , i.e.,  $\forall (\ell, q) : |(\theta_{\ell,q}(\mathbf{k}))| \geq \gamma$ . In fact, each element of vector  $\boldsymbol{E}[I_1]$  in (33) is positive if either the sign of its corresponding term in vector  $\kappa \tau B^\top B \boldsymbol{E}[\boldsymbol{\alpha}(\mathbf{k})] \Psi(\mathbf{k})$  (31) is negative or the sign is positive and the size is smaller than  $\gamma$ . Hence,  $\gamma > \kappa \tau (d_{\max} + 1) \mu_M \Psi_{\max}$  should hold, where  $d_{\max}$  is the maximum degree of the nodes of

the underlying graph. Since  $\lambda_{\max}(L_e) \ge (d_{\max} + 1)$ , the bound in (42) gives a smaller  $\tau$ , hence, it satisfies the required condition. Thus,  $|\Psi^{\top}(\mathbf{k})||\mathbf{E}[I_1]| = \Psi^{\top}(\mathbf{k})\mathbf{E}[I_1]$ . We now write  $\Delta V_M$  and argue similar to the case of  $V_m$  which gives (16).

The sufficient coupling condition  $\kappa$  in the above result depends on  $|\Psi(\gamma)|$ . For relative phases belonging to  $\Upsilon$ ,  $|\Psi(\gamma)|$  is the minimum of  $\Psi(\gamma_i), \gamma_i \in \Upsilon$ . That is, the obtained condition guarantees that the chain will return to  $S_G^{\Psi}(\gamma)$  from  $\Upsilon$ . We notice that even if the chain initiates from  $S_G^{\Psi}(\gamma)$ , the presence of stochastic uncertainties can transfer the relative phase of each two oscillators from  $S_G^{\Psi}(\gamma)$  to not only  $\Upsilon$  but also  $\overline{\Upsilon}$ . Considering returning of the relative phases from  $\overline{\Upsilon}$  to  $S_G^{\Psi}(\gamma)$ , we shall replace  $|\Psi(\gamma)|$  in (16) with  $\Psi(\gamma_{\min}) = \min \Psi(\gamma_i), \gamma_i \in \overline{\Upsilon}$ . This substitution leads to  $\kappa \to \infty$  (see Example 1). To verify whether the arc set  $\overline{\Upsilon}$  is absorbing, we shall study the evolution of the relative phase after exiting  $\overline{\Upsilon}$ . We can prove that if  $E[\alpha] > 0$ , the chain is transient with respect to  $S_G^{\Psi}(\gamma_{\max})$ . Recall that similar to  $\underline{\Upsilon}$ , the maximum value of  $\Psi(\cdot)$  for the arcs in  $\overline{\Upsilon}$  is  $|\Psi(\gamma)|$ .

$$\bar{V}(\Theta(\mathbf{k})) = C^{\top}(\pi \mathbf{1}_m - |B^{\top} \boldsymbol{\theta}(\mathbf{k})|).$$

Computing the one-step drift, we obtain

$$\Delta \bar{V}(\Theta) = -\Delta V(\Theta). \tag{43}$$

Therefore, assuming  $E[\alpha]$  is positive definite, the lower bound for  $\kappa$  in (16), which guarantees  $\Delta V < 0$ , leads to  $\Delta \overline{V} > 0$ . Based on [29, Th. 8.0.2], the chain is transient on  $S_G^{\Psi}(\gamma_{\max})$ . That is, the probability that the chain revisits  $S_G^{\Psi}(\gamma_{\max})$  infinitely often is zero. From the analyses using both V and  $\overline{V}$ , the conditions in (16) guarantee that the chain is stochastic phase-cohesive with respect to  $S_G^{\Psi}(\gamma)$ .

Now, assume that  $\boldsymbol{E}[\boldsymbol{\alpha}]$  is negative definite, and the underlying deterministic topology satisfies  $B^{\top}\boldsymbol{\theta} \geq \gamma_{\max}\mathbf{1}_m \in \mathcal{R}(B^{\top})$ . The latter assumption is imposed dealing with the existence of graph cycles (see [50]) composed of an odd number of oscillators. In such a case, independent of the stochastic nature of our problem setting, it is not feasible to have all relative phases greater than some predefined limits, e.g., if  $\gamma_{\max} > \frac{\pi}{2}$ . Hence, at least one of the relative phases should be confined to the arc  $\underline{\Upsilon}$  by the topological restrictions. Here, we exempt the latter case. To have  $\Delta \bar{V}(\Theta) < 0$ ,  $C^{\top}|B^{\top}\boldsymbol{\theta}(\mathbf{k})| \geq C^{\top}\boldsymbol{E}[|B^{\top}\boldsymbol{\theta}(\mathbf{k}+1)|]$ should hold. Since  $\boldsymbol{E}[|B^{\top}\boldsymbol{\theta}(\mathbf{k}+1)|] \geq |\boldsymbol{E}[B^{\top}\boldsymbol{\theta}(\mathbf{k}+1)]|$ elementwise, therefore we should have

$$C^{\top}|B^{\top}\boldsymbol{\theta}(\mathbf{k})| \ge C^{\top}|\boldsymbol{E}[B^{\top}\boldsymbol{\theta}(\mathbf{k}+1)]|.$$
(44)

Similar to the previous case, we study two cases of one edge  $\bar{V}_m$  and all edges  $\bar{V}_M$  belonging to  $\Upsilon$ . Considering the case of  $\bar{V}_m$ , we assume  $|(\theta_{\ell,q}(\mathbf{k}))| \leq \gamma_{\max}$  and  $|\theta_{p,s}(\mathbf{k})| = \pi \forall (p,s) \neq (\ell,q)$ . Since  $\Psi(\pi) = 0$ , we can replace C with  $|\Psi^{\top}(\mathbf{k})|$  in both sides of (44). As a result, the following should hold:

$$|\underbrace{\Psi^{\top}(\mathbf{k})B^{\top}\boldsymbol{\theta}(\mathbf{k})}_{a} + \underbrace{\tau\kappa \ \Psi^{\top}(\mathbf{k})B^{\top}B|\boldsymbol{E}[\boldsymbol{\alpha}(\mathbf{k})]|\Psi(\mathbf{k})}_{b} + \underbrace{\tau\Psi^{\top}(\mathbf{k})\boldsymbol{E}[B^{\top}\boldsymbol{\tilde{\omega}}(\mathbf{k})]}_{c}| \geq \Psi^{\top}(\mathbf{k})B^{\top}\boldsymbol{\theta}(\mathbf{k}).$$
(45)

Since  $\Psi(\cdot)$  is an odd function, b > 0 holds (as discussed above). Then, if  $a + b \pm |c| > 0$ , to have a negative drift,  $\min\{b\} > \max\{|c|\}$  should hold, which gives the condition on  $\kappa$ , as in (17). If the latter condition holds,  $a + b \pm |c|$  is always positive. However, the relative phase at each time should be smaller than  $\pi$  (definition of geodesic distance), thus  $\max\{b + c\} < \Psi_{\max}(\pi - \gamma_{\max})$  should hold, which completes the proof.  $\Box$ 

#### APPENDIX E PROOF OF COROLLARY 1

*Proof:* The proof is based on [29, Th. 11.0.1]. We derive the conditions under which the following inequality holds:

$$\boldsymbol{E}[V(\Phi_{k+1})|\Phi_k = x] - V(x) < -1 \quad \forall \Phi_k \in \Pi \setminus S_G^{\Psi}(\gamma).$$
(46)

The rest of the proof is similar to the proof of Theorem 1.  $\Box$ 

# APPENDIX F PROOF OF PROPOSITION1

**Proof:** Take  $\bar{V} = \max_{i,j} |\Psi(\theta_{i,j})|$  and assume that at time k,  $\{\exists(\ell, e) \in \mathcal{E} : |\theta_{\ell,e}| \in \Upsilon, \bar{V}(k) = |\Psi(\theta_{\ell,e})|\}$ . To prove the stochastic phase-cohesiveness w.r.t.  $U_G^{\Psi}(\gamma)$ , we use a drift-based argument to derive conditions under which  $\Delta \bar{V} = \boldsymbol{E}[\bar{V}(k+1)] - |\Psi(\theta_{\ell,e})(k)|$  is negative. Recall that from the definition of the desired sets in Section III-B,  $\Psi(\cdot)$  for all arcs that belong to  $\Upsilon$  is larger than the arcs in  $\underline{\Upsilon}$  or  $\overline{\Upsilon}$ . So, instead of computing  $\Psi(\theta_{\ell,e}(k+1))$ , we use an equivalent argument. We compute the evolution of edge (relative phase)  $\theta_{\ell,e}$ , that is,  $\Delta \theta_{\ell,e} = \boldsymbol{E}[|\theta_{\ell,e}(k+1)|] - |\theta_{\ell,e}(k)|$ . We prove that an edge for which  $\operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}]) > 0$  holds is recurrent to  $\underline{\Upsilon}$  while if  $\operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}]) < 0$  holds, it is recurrent to  $\underline{\Upsilon}$ . In both cases,  $\Delta \bar{V} < 0$  is guaranteed. Assume  $\theta_{\ell,e}(k+1)]$ . Since the underlying topology is a line graph, based on (12), we have

$$\Delta \theta_{\ell,e} = \boldsymbol{E}[\theta_{\ell,e}(\mathbf{k}+1)] - \theta_{\ell,e}(\mathbf{k})$$
  
=  $-2 \tau \kappa \operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}(\mathbf{k})]) | \boldsymbol{E}[\tilde{\alpha}_{\ell,e}(\mathbf{k})] | \Psi(\theta_{\ell,e}(\mathbf{k}))$   
 $- \tau \kappa \operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{p,q}(\mathbf{k})]) | \boldsymbol{E}[\tilde{\alpha}_{p,q}(\mathbf{k})] | \Psi(\theta_{p,q}(\mathbf{k}))$   
 $- \tau \kappa \operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{d,s}(\mathbf{k})]) | \boldsymbol{E}[\tilde{\alpha}_{d,s}(\mathbf{k})] | \Psi(\theta_{d,s}(\mathbf{k}))$  (47)

where (p,q) and (d,s) denote the neighboring edges. Our aim is to prove that  $\operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}(\mathbf{k})])\Delta\theta_{\ell,e} < 0$ . Recall that  $|\Psi(\theta_{\ell,e}(\mathbf{k}))| \geq \Psi(\gamma)$ , where  $|\Psi(\theta_{\ell,e}(\mathbf{k}))|$  is the maximum. That is,  $|\Psi(\theta_{d,s}(\mathbf{k}))| \leq |\Psi(\theta_{\ell,e}(\mathbf{k}))|$  and  $|\Psi(\theta_{p,q}(\mathbf{k}))| \leq |\Psi(\theta_{\ell,e}(\mathbf{k}))|$ . Since the size of all mean-values are equal to  $\lambda$ and  $\Psi(\theta_{\ell,e}(\mathbf{k}))|$ . Since the sign of  $\Delta\theta_{\ell,e}$  is always equal to  $-\operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}(\mathbf{k})])$  unless  $|\Delta\theta_{\ell,e}| = 0$ . In fact, depending on the sign and size of  $\Psi(\theta_{p,q}(\mathbf{k}))$  and  $\Psi(\theta_{d,s}(\mathbf{k}))$ , we have

$$0 \le |\Delta \theta_{\ell,e}| \le 4\tau \kappa \lambda \Psi_{\max}.$$

Let us look at the case of  $|\Delta \theta_{\ell,e}| = 0$ . In this case, the  $\Psi(\cdot)$  of the neighboring edges are equal to  $\Psi(\theta_{\ell,e}(\mathbf{k}))$ . We consider the dynamics of either of them and write their evolution similar to (47). Now, either the same situation occurs or the maximum relative phase is dominant w.r.t. its neighbors. In case of a zero difference, we continue with a neighboring edge and repeat this process till

Authorized licensed use limited to: TU Delft Library. Downloaded on April 26,2024 at 13:49:02 UTC from IEEE Xplore. Restrictions apply.

reaching the tail of the graph or if we find a maximum edge whose neighbors (at least one neighbor) do not possess a corresponding maximum  $\Psi(\cdot)$ . Since the graph is a line graph, the final edge of the graph has only one neighbor. Then we can write,

$$\begin{aligned} \Delta \theta_{\ell,e} &= \boldsymbol{E}[\theta_{\ell,e}(\mathbf{k}+1)] - \theta_{\ell,e}(\mathbf{k}) \\ &= -2 \,\tau \,\kappa \, \mathrm{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}(\mathbf{k})]) |\boldsymbol{E}[\tilde{\alpha}_{\ell,e}(\mathbf{k})] |\Psi(\theta_{\ell,e}(\mathbf{k})) \\ &- \tau \,\kappa \, \mathrm{sign}(\boldsymbol{E}[\tilde{\alpha}_{p,q}(\mathbf{k})]) |\boldsymbol{E}[\tilde{\alpha}_{p,q}(\mathbf{k})] |\Psi(\theta_{p,q}(\mathbf{k})). \end{aligned}$$

$$(48)$$

From the above we conclude that for  $\kappa > 0$ ,  $|\Delta \theta_{\ell,e}| \ge \tau \kappa \lambda \Psi_{\gamma} \ne 0$  and  $\operatorname{sign}(\Delta \theta_{\ell,e}) = -\operatorname{sign}(\boldsymbol{E}[\tilde{\alpha}_{\ell,e}])$ . This gives  $\Delta \bar{V} < 0$ . Now, consider the case of  $\boldsymbol{E}[|\theta_{\ell,e}(\mathbf{k}+1)|] = -\boldsymbol{E}[\theta_{\ell,e}(\mathbf{k}+1)]$ . Writing the inequality in (47) for the maximum case, we obtain the condition on the sampling time, i.e.,  $-\gamma + 4\tau\kappa\lambda\Psi_{\max} < \gamma$  which completes the proof.

# APPENDIX G PROOF OF PROPOSITION 2

*Proof:* Similar to the proof of Theorem 1, define

$$\begin{split} V(\Theta(\mathbf{k})) &= |\Psi_r(\gamma)| \sum_{\substack{|\theta_{i,j}(\mathbf{k})| \in \Upsilon}} \mu_{i,j} |\theta_{i,j}(\mathbf{k})| \\ &+ \Psi^o \sum_{\substack{|\theta_{i,j}(\mathbf{k})| \in \underline{\Upsilon}^1 \cup \underline{\Upsilon}^2}} \mu_{i,j} |\theta_{i,j}(\mathbf{k})| \end{split}$$

where  $\mu_{i,j} = E[\tilde{\alpha}_{i,j}]$ . Assume that at time k, there exists an edge  $\theta_{\ell,e}$  such that  $|\theta_{\ell,e}| \in \Upsilon$ . To characterize the lower bound on  $\kappa$ , we assume the worst condition, i.e.,  $|(\theta_{\ell,q}(\mathbf{k}))| = \gamma$  and  $|\theta_{p,s}(\mathbf{k})| \leq \gamma_c \forall (p,s) \neq (\ell,q)$  and  $\Psi(\theta_{p,s}(\mathbf{k})) = -\overline{\Psi}$ . Similar to the proof of Theorem 1, it holds  $V(\Theta(\mathbf{k}+1)) \leq \overline{V}(\Theta(\mathbf{k}+1))$ , where

$$\bar{V}(\Theta(\mathbf{k}+1)) = |\Psi_{r,\mu}^{\top}(B^{\top}\boldsymbol{\theta}(\mathbf{k}))||B^{\top}\boldsymbol{\theta}(\mathbf{k}+1)|$$

with  $\Psi_{r,\mu}(B^{\top}\boldsymbol{\theta}(\mathbf{k})) = \mu\Psi_r(B^{\top}\boldsymbol{\theta}(\mathbf{k}))$ , and  $\mu_{m\times m}$  is a constant matrix equal to  $\mu = \boldsymbol{E}[\boldsymbol{\alpha}]$ . Different from Theorem 1, there exists an arc in  $[0, \pi]$  on which  $\Psi_r(\cdot)$  may take positive or negative values. Hence,  $\Psi_{r,\mu}^{\top}(B^{\top}\boldsymbol{\theta}(\mathbf{k}))B^{\top}\boldsymbol{\theta}(\mathbf{k}) \geq 0$  does not necessarily hold. Denote  $\bar{V}(\Theta(\mathbf{k})), \Psi_{r,\mu}^{\top}(B^{\top}\boldsymbol{\theta}(\mathbf{k}))$ , and  $\Psi_r(B^{\top}\boldsymbol{\theta}(\mathbf{k}))$  by  $\bar{V}(\mathbf{k}), \Psi_{r,\mu}^{\top}(\mathbf{k})$ , and  $\Psi_r(\mathbf{k})$ , respectively. Since, the variances are assumed to be small, we have

$$\Delta V(\Theta) \leq -V(\mathbf{k}) + E[\underbrace{\tau | \Psi_{r,\mu}^{+}(\mathbf{k})| | B^{\top} \tilde{\boldsymbol{\omega}}(\mathbf{k})}_{c} | \boldsymbol{\theta}(\mathbf{k})] + E[\underbrace{| \Psi_{r,\mu}^{+}(\mathbf{k})| B^{\top} \boldsymbol{\theta}(\mathbf{k})}_{a} - \tau \kappa \underbrace{| \Psi_{r,\mu}^{+}(\mathbf{k})| B^{\top} B \underline{\boldsymbol{\alpha}}(\mathbf{k}) \Psi_{r}(\mathbf{k})}_{b} | \boldsymbol{\theta}(\mathbf{k})].$$
(49)

Similar to the proof of Theorem 1, to have  $\Delta \overline{V} < 0$ , the inequalities in (36) should hold. Now, we continue by characterizing the lower bound on  $\kappa$ , we assume the worst condition, i.e.,  $|(\theta_{\ell,q}(\mathbf{k}))| = \gamma$  and  $|\theta_{p,s}(\mathbf{k})| < \gamma_c \forall (p,s) \neq (\ell,q)$  and  $\Psi(\theta_{p,s}(\mathbf{k})) = -\overline{\Psi}$ . Considering the worst condition, assume that  $E[\tilde{\alpha}_{\ell,q}] = \lambda_{\min}(E[\alpha])$  and  $E[\tilde{\alpha}_{p,s}] = \lambda_{\max}(E[\alpha])$ . From the definition of  $\overline{V}(\boldsymbol{\theta}(\mathbf{k}))$ , we obtain  $a = \overline{V}(\mathbf{k})$ . Define,

$$\begin{aligned} x &= \lambda_{\min}(\boldsymbol{E}[\boldsymbol{\alpha}])[|\Psi_r(\boldsymbol{\gamma})| \ 0 \dots 0]^\top \\ y &= \lambda_{\max}(\boldsymbol{E}[\boldsymbol{\alpha}])[0 \ \bar{\Psi} \ \bar{\Psi} \dots \bar{\Psi}]^\top. \end{aligned}$$

Then, we can write  $|\Psi_{r,\mu}(\mathbf{k})| = x + y$  and  $E[\alpha(\mathbf{k})]\Psi_r(\mathbf{k}) = x - y$ . Thus, computing the term b in (49) gives

$$\boldsymbol{E}[b] = (\boldsymbol{x}^{\top} + \boldsymbol{y}^{\top})\boldsymbol{B}^{\top}\boldsymbol{B}(\boldsymbol{x} - \boldsymbol{y}) = \boldsymbol{x}^{\top}\boldsymbol{B}^{\top}\boldsymbol{B}\boldsymbol{x} - \boldsymbol{y}^{\top}\boldsymbol{B}^{\top}\boldsymbol{B}\boldsymbol{y}.$$

Notice that  $B^{+}B$  is symmetric, and for a tree graph  $L_{e} = B^{+}B$  is positive definite. Define  $\hat{\lambda}_{1} = \lambda_{\min}(\boldsymbol{E}[\alpha])|\Psi_{r}(\gamma)|$  and  $\hat{\lambda}_{2} = \lambda_{\max}(\boldsymbol{E}[\alpha])\bar{\Psi}$ . Assume that

$$\hat{\lambda} = \hat{\lambda}_1 - \hat{\lambda}_2 \sqrt{(m-1)} > 0.$$

Denote  $f = \hat{\lambda}_1 + \hat{\lambda}_2 \sqrt{(m-1)} > 0$ . As a result,  $E[b] \ge \lambda_{\min}(L_e)\hat{\lambda}f > 0$ . To obtain  $\kappa$ , we should have  $\kappa \min \{E[b]\} > \max E[c]$ . This gives,

$$\kappa > \frac{\left(\hat{\lambda}_1 + \hat{\lambda}_2(m-1)\right) \boldsymbol{E}_{\max}[|\Delta \tilde{\omega}|]}{\lambda_{\min}(L_e)\hat{\lambda}\left(\hat{\lambda}_1 + \hat{\lambda}_2\sqrt{(m-1)}\right)}$$

To obtain the bound on  $\tau$ , we assume that all edges of the network belong to  $\Upsilon$ . The rest of the proof follows from the proof of Theorem 1 which leads to the conditions in (23).

# APPENDIX H PROOF OF LEMMA 3

*Proof:* The network topology is a random graph, therefore, at each time step k, the set of randomly established edges determines the space wherein the relative phases evolve until k + 1. Thus, the probability space is the countable set of spaces that are randomly selected by multiple independent Bernoulli processes. The probability space may contain at least one space (correspond to the null graph) and at most  $2^m$  spaces (corresponding to the maximal graph). Since p is a fixed nonzero probability and the topology of the network at a time k is independent of the states  $\theta_i(t), t < k$ , the dynamics  $\theta_i(k+1)$  only depends on  $\theta_i(k)$ , and the random topology determined by p and hence (24) is a Markov chain. Due to the independence of the Bernoulli processes over time and also p being a nonzero probability, the countable set of probability spaces is  $\psi$ -irreducible (Definition 3, also [29, Ch.4]).  $\square$ 

#### APPENDIX I PROOF OF THEOREM 2

**Proof:** The proof is based on [29, Th. 11.0.1] for chains evolving in a countable space. Notice that here, all compact sets are petite since the space is countable. The trend of the proof is similar to the proof of Theorem 1. Define function  $V(\Theta(\mathbf{k}))$ , as in (29). Different from Theorem 1, here, the randomness is governed by the Bernoulli distribution and we can compute the finite set of all possible outcomes. In order to compute the one-step drift for  $V_m$  and  $V_M$  (as defined in the proof of Theorem 1), we need to calculate  $\mathbf{E}[|I_1|]$ , with  $I_1 = B^{\top} \boldsymbol{\theta}(\mathbf{k}) - \tau \kappa B^{\top} B \beta(\mathbf{k}) \Psi(B^{\top} \boldsymbol{\theta}(\mathbf{k}))$ . Since each of the two oscillators, if connected, are linked undirectedly, we can write B such that  $B^{\top} \theta(\mathbf{k}) \succ 0$  holds. Recall that based on the definition of  $V_m$ , only one edge is nonzero, and for  $V_M$ , all are nonzero. Now, consider  $I_1$ , also in the view of dynamics in (26). If  $\tau$  is sufficiently small, for a given  $\kappa$ , we can assure that all elements of  $I_1$  are positive, hence,  $\mathbf{E}[|I_1|] = \mathbf{E}[I_1]$  holds. To characterize the condition on  $\tau$ , we write

$$\gamma = \min |\theta_{i,j}(\mathbf{k})| \ge \tau \kappa \, d_{\max} \Psi_{\max}$$

where  $d_{\max}$  is the maximum degree, and  $d_{\max} \leq \lambda_{\max}(L_e)$ . This gives the bound on  $\tau$ . Then, based on a similar argument as the proof of Theorem 1, we show that the one-step drift of Vfrom  $\Upsilon$  in (6) to  $\underline{\Upsilon}$  in (5) is negative if  $\kappa$  is sufficiently large, as in (27). Furthermore, since  $\forall i, j, \omega_{i,j} \neq 0$  and the oscillators are randomly connected, if the maximum relative phase enters  $\overline{\Upsilon}$ , the probability that it exits this set is nonzero. In a similar fashion to Example 1 and Theorem 1, we argue that  $S_G^{\Psi}(\gamma_{\max})$ is transient which ends the proof.  $\Box$ 

# APPENDIX J PROOF OF COROLLARY 2

*Proof:* The proof follows a similar trend as of Theorem 2. Set  $\gamma = 0^+$  which gives  $S_G^{\Psi}(0) = \{\theta_i, \theta_j \in \mathbb{S}^1 : |\theta_{i,j}(\mathbf{k})| = \gamma \forall (i, j) \in \mathcal{E}\}$ . From the result of Theorem 2, we have

$$\kappa \tau |\Psi(\gamma)| p \lambda_{\min}(L_e) > |\Delta_{\max}\omega|.$$
(50)

Notice that the above holds for any  $\gamma \in [0^+, \pi^-]$ . By substituting  $|\Delta_{\max}\omega| = 0$ , we conclude that the condition is satisfied  $\forall \kappa > 0$ . From the proof of Theorem 2, we have  $\tau \sim \frac{\gamma}{\kappa}$ , and hence sufficiently small. This indicates that if the chain exits the origin, it revisits the origin with probability one. Notice that this does not hold for the antiphase arc  $\overline{\Upsilon} = \pi$ . Since, exiting this arc, the relative phases will return to the origin, as explained above. This completes the proof.

#### REFERENCES

- [1] E. Izhikevich, *Dynamical Systems in Neuroscience*. Cambridge, MA, USA: MIT Press, 2007.
- [2] R. Sepulchre, "Oscillators as systems and synchrony as a design principle," in *Current Trends in Nonlinear Systems and Control*. Berlin, Germany: Springer, 2006, pp. 123–141.
- [3] S. Strogatz, "From Kuramoto to Crawford: Exploring the onset of synchronization in populations of coupled oscillators," *Physica D, Nonlinear Phenomena*, vol. 143, no. 1/4, pp. 1–20, 2000.
- [4] M. Jafarian and K. H. Johansson, "Synchronization of quadratic integrateand-fire spiking neurons: Constant versus voltage-dependent couplings," in *Proc. IEEE 58th Conf. Decis. Control*, 2019, pp. 4711–4716.
- [5] E. Nozari and J. Cortés, "Oscillations and coupling in interconnections of two-dimensional brain networks," in *Proc. IEEE Amer. Control Conf.*, 2019, pp. 193–198.
- [6] G. Villani, M. Jafarian, A. Lansner, and K. H. Johansson, "Analysis of free recall dynamics of an abstract working memory model," in *Proc. IEEE Amer. Control Conf.*, 2020, pp. 2562–2567.
- [7] A. Ijspeert, J. Nakanishi, H. Hoffmann, P. Pastor, and S. Schaal, "Dynamical movement primitives: Learning attractor models for motor behaviors," *Neural Comput.*, vol. 25, no. 2, pp. 328–373, 2013.
- [8] C. Börgers and N. Kopell, "Synchronization in networks of excitatory and inhibitory neurons with sparse, random connectivity," *Neural Comput.*, vol. 15, no. 3, pp. 509–538, 2003.
- [9] G. Deco, V. Jirsa, A. McIntosh, O. Sporns, and R. Kötter, "Key role of coupling, delay, and noise in resting brain fluctuations," *Proc. Nat. Acad. Sci.*, vol. 106, no. 25, pp. 10302–10307, 2009.

- [10] M. Richardson and W. Gerstner, "Synaptic shot noise and conductance fluctuations affect the membrane voltage with equal significance," *Neural Comput.*, vol. 17, no. 4, pp. 923–947, 2005.
- [11] J. White, J. Rubinstein, and A. Kay, "Channel noise in neurons," *Trends Neurosciences*, vol. 23, no. 3, pp. 131–137, 2000.
- [12] H. Khalil, Nonlinear Systems. Englewood Cliffs, NJ, USA: Prentice Hall, 2002.
- [13] E. Izhikevich and Y. Kuramoto, "Weakly coupled oscillators," *Encyclopedia Math. Phys.*, vol. 5, 2006, Art. no. 448.
- [14] P. Sacre and R. Sepulchre, "Sensitivity analysis of oscillator models in the space of phase-response curves: Oscillators as open systems," *IEEE Control Syst. Mag.*, vol. 34, no. 2, pp. 50–74, Feb. 2014.
- [15] E. Brown, P. Holmes, and J. Moehlis, "Globally coupled oscillator net-works," in *Perspectives and Problems in Nolinear Science*. Berlin, Germany: Springer, 2003, pp. 183–215.
  [16] E. Izhikevich, "Weakly pulse-coupled oscillators, FM interactions, syn-
- [16] E. Izhikevich, "Weakly pulse-coupled oscillators, FM interactions, synchronization, and oscillatory associative memory," *IEEE Trans. Neural Netw.*, vol. 10, no. 3, pp. 508–526, May 1999.
- [17] F. Dörfler and F. Bullo, "Synchronization in complex networks of phase oscillators: A survey," *Automatica*, vol. 50, no. 6, pp. 1539–1564, 2014.
- [18] A. Jadbabaie, N. Motee, and M. Barahona, "On the stability of the Kuramoto model of coupled nonlinear oscillators," in *Proc. Amer. Control Conf.*, 2004, pp. 4296–4301.
- [19] R. Delabays, P. Jacquod, and F. Dörfler, "The Kuramoto model on oriented and signed graphs," *SIAM J. Appl. Dynamical Syst.*, vol. 18, no. 1, pp. 458–480, 2019.
- [20] A. Franci, A. Chaillet, and W. Pasillas-Lépine, "Existence and robustness of phase-locking in coupled Kuramoto oscillators under mean-field feedback," *Automatica*, vol. 47, no. 6, pp. 1193–1202, 2011.
- [21] M. Jafarian, X. Yi, M. Pirani, H. Sandberg, and K. H. Johansson, "Synchronization of Kuramoto oscillators in a bidirectional frequency-dependent tree network," in *Proc. IEEE 57th Conf. Decis. Control*, 2018, pp. 4505– 4510.
- [22] T. Menara, G. Baggio, D. Bassett, and F. Pasqualetti, "Stability conditions for cluster synchronization in networks of heterogeneous Kuramoto oscillators," *IEEE Trans. Control Netw. Syst.*, vol. 7, no. 1, pp. 302–314, Mar. 2020.
- [23] K. Okuda, "Variety and generality of clustering in globally coupled oscillators," *Physica D, Nonlinear Phenomena*, vol. 63, no. 3/4, pp. 424–436, 1993.
- [24] S. Watanabe and J. Swift, "Stability of periodic solutions in series arrays of Josephson junctions with internal capacitance," *J. Nonlinear Sci.*, vol. 7, no. 6, pp. 503–536, 1997.
- [25] T. Nishikawa, F. Hoppensteadt, and Y. Lai, "Oscillatory associative memory network with perfect retrieval," *Physica D, Nonlinear Phenomena*, vol. 197, no. 1/2, pp. 134–148, 2004.
- [26] J. Acebrón, L. Bonilla, C. Vicente, F. Ritort, and R. Spigler, "The Kuramoto model: A simple paradigm for synchronization phenomena," *Rev. Modern Phys.*, vol. 77, no. 1, 2005, Art. no. 137.
- [27] E. Mallada and A. Tang, "Synchronization of weakly coupled oscillators: Coupling, delay and topology," *J. Phys. A, Math. Theor.*, vol. 46, no. 50, 2013, Art. no. 505101.
- [28] H. Kushner, "Numerical methods for stochastic control problems in continuous time," SIAM J. Control Optim., vol. 28, no. 5, pp. 999–1048, 1990.
- [29] S. Meyn and R. Tweedie, Markov Chains and Stochastic Stability. Berlin, Germany: Springer, 2012.
- [30] J. Teramae and D. Tanaka, "Robustness of the noise-induced phase synchronization in a general class of limit cycle oscillators," *Phys. Rev. Lett.*, vol. 93, no. 20, 2004, Art. no. 204103.
- [31] B. Bag, K. Petrosyan, and C. Hu, "Influence of noise on the synchronization of the stochastic Kuramoto model," *Phys. Rev. E*, vol. 76, no. 5, 2007, Art. no. 056210.
- [32] H. Yin, P. Mehta, S. Meyn, and U. Shanbhag, "Synchronization of coupled oscillators is a game," *IEEE Trans. Autom. Control*, vol. 57, no. 4, pp. 920–935, Apr. 2012.
- [33] T. Ichinomiya, "Frequency synchronization in a random oscillator network," *Phys. Rev. E*, vol. 70, no. 2, 2004, Art. no. 026116.
- [34] P. Erdös and A. Rényi, "On random graphs," France: Publicationes Mathematicae, vol. 6, pp. 290–297, 1995.
- [35] Y. Hatano and M. Mesbahi, "Agreement over random networks," *IEEE Trans. Autom. Control*, vol. 50, no. 11, pp. 1867–1872, Nov. 2005.
- [36] S. Manaffam and A. Seyedi, "Synchronization probability in large complex networks," *IEEE Trans. Circuits Syst. II, Exp. Briefs*, vol. 60, no. 10, pp. 697–701, Oct. 2013.

- [37] V. Preciado and G. Verghese, "Synchronization in generalized Erdös-Rényi networks of nonlinear oscillators," in *Proc. IEEE 44th Conf. Decis. Control*, 2005, pp. 4628–4633.
- [38] M. Porfiri, D. Stilwell, and E. Bollt, "Synchronization in random weighted directed networks," *IEEE Trans. Circuits Syst. I, Regular Papers*, vol. 55, no. 10, pp. 3170–3177, Nov. 2008.
- [39] C. Zhou, A. Motter, and J. Kurths, "Universality in the synchronization of weighted random networks," *Phys. Rev. Lett.*, vol. 96, no. 3, 2006, Art. no. 034101.
- [40] J. Wang, C. Xu, M. Chen, J. Feng, and G. Chen, "Stochastic feedback coupling synchronization of networked harmonic oscillators," *Automatica*, vol. 87, pp. 404–411, 2018.
- [41] Z. Aminzare and V. Srivastava, "Stochastic synchronization in nonlinear network systems driven by intrinsic and coupling noise," *Biol. Cybern.*, vol. 116, no. 2, pp. 147–162, 2022.
- [42] G. Russo, F. Wirth, and R. Shorten, "On synchronization in continuoustime networks of nonlinear nodes with state-dependent and degenerate noise diffusion," *IEEE Trans. Autom. Control*, vol. 64, no. 1, pp. 389–395, Jan. 2019.
- [43] A. Franci, A. Chaillet, and W. Pasillas-Lépine, "Phase-locking between Kuramoto oscillators: Robustness to time-varying natural frequencies," in *Proc. IEEE 49th Conf. Decis. Control*, 2010, pp. 1587–1592.
- [44] W. Lu and F. Atay, "Stability of phase difference trajectories of networks of Kuramoto oscillators with time-varying couplings and intrinsic frequencies," *SIAM J. Appl. Dynamical Syst.*, vol. 17, no. 1, pp. 457–483, 2018.
- [45] L. Zhu and D. Hill, "Synchronization of Kuramoto oscillators: A regional stability framework," *IEEE Trans. Autom. Control*, vol. 65, no. 12, pp. 5070–5082, Dec. 2020.
- [46] C. Favaretto, A. Cenedese, and F. Pasqualetti, "Cluster synchronization in networks of Kuramoto oscillators," *IFAC-PapersOnLine*, vol. 50, no. 1, pp. 2433–2438, 2017.
- [47] D. Klein, P. Lee, K. Morgansen, and T. Javidi, "Integration of communication and control using discrete time Kuramoto models for multivehicle coordination over broadcast networks," *IEEE J. Sel. Areas Commun.*, vol. 26, no. 4, May 2008.
- [48] J. Willems and J. Willems, "Feedback stabilizability for stochastic systems with state and control dependent noise," *Automatica*, vol. 12, no. 3, pp. 277–283, 1976.
- [49] M. Jafarian, M. H. Mamduhi, and K. H. Johansson, "Stochastic phasecohesiveness of discrete-time Kuramoto oscillators in a frequencydependent tree network," in *Proc. IEEE 18th Eur. Control Conf.*, 2019, pp. 1987–1992.
- [50] M. Mesbahi and M. Egerstedt, Graph Theoretic Methods in Multiagent Networks. Princeton, NJ, USA: Princeton Univ. Press, 2010.
- [51] S. Meyn and R. L. Tweedie, "State-dependent criteria for convergence of Markov chains," Ann. Appl. Probability, vol. 4, no. 1, pp. 149–168, 1994.
- [52] F. Leone, L. Nelson, and R. Nottingham, "The folded normal distribution," *Technometrics*, vol. 3, no. 4, pp. 543–550, 1961.



Matin Jafarian (Member, IEEE) received the B.Sc. degree in electrical engineering, majoring in systems and control from the K. N. Toosi University of Technology, Tehran, Iran, in 2003, the M.Sc. degree in electrical engineering, majoring in systems and control from the University of Twente, The Netherlands, in 2010, and the Ph.D. degree in systems and control theory from ENTEG Institute, University of Groningen, Groningen, Netherlands, in 2015.

She has held a postdoctoral position with the Division of Decision and Control Systems, KTH Royal institute of Technology, Stockholm, Sweden. She is currently an Assistant Professor with the Delft Center for Systems and Control, TU Delft, Delft, Netherlands, and a Research fellow with the Donders Institute for Brain, Cognition, and Behavior, Nijmegen, The Netherlands. She was a Marie Skłodowska-Curie fellow jointly with TU Delft and Donders Institute. Her research interests include analysis and control of nonlinear and stochastic dynamic networks.



**Mohammad H. Mamduhi** (Senior Member, IEEE) received the B.Sc. degree in mechanical engineering from the Sharif University of Technology, Tehran, Iran, in 2008, the master's degree in systems, control, and robotics from the KTH Royal institute of Technology (KTH), Stockholm, Sweden, in 2010, and the Ph.D. degree in electrical engineering and computer science from the Technical University of Munich, Munich, Germany, in 2017.

He was a Postdoctoral Researcher with the Division of Decision and Control Systems, KTH. He was a Senior Scientist with the Automatic Control Laboratory, Department of Information Technology and Electrical Engineering, ETH Zürich, Zürich, Switzerland. His research interests include networked control systems, cyberphysical systems, and stochastic systems.



Karl H. Johansson (Fellow, IEEE) received the M.Sc. degree in electrical engineering and the Ph.D. degree in automatic control from Lund University, Lund, Sweden, in 1992 and 1997, respectively.

He has held visiting positions with UC Berkeley, Berkeley, CA, USA; Caltech, Pasadena, CA, USA; and NTU, Singapore, among other institutions. He is currently a Swedish Research Council Distinguished Professor of Electrical Engineering and Computer Science with the

KTH Royal Institute of Technology, Stockholm, Sweden, where he is also the Director of Digital Futures. He has coauthored more than 800 journal and conference papers in the ares of his research interests, and supervised almost 100 postdocs and Ph.D. students. His research interests include networked control systems and cyber-physical systems with applications in transportation, energy, and automation networks.

Prof. Johansson is a Fellow the Royal Swedish Academy of Engineering Sciences. He was the recipient of several best paper awards and other distinctions from IEEE, IFAC, and ACM. He was the recipient of Distinguished Professor by Swedish Research Council, Wallenberg Scholar with the Knut and Alice Wallenberg Foundation, Future Research Leader from the Swedish Foundation for Strategic Research, the triennial IFAC Young Author Prize, and IEEE Control Systems Society Distinguished Lecturer. He has served on the IEEE Control Systems Society Board of Governors and the Swedish Scientific Council for Natural Sciences and Engineering Sciences. He is the President of the European Control Association and member of the IFAC Council.